EXHIBIT A

Existing Affected Tariff Pages

5.0 NETWORK SERVICES DESCRIPTIONS

5.1 Service Offerings

The following Network Services for residence/business customers and for carriers certificated by the Public Utilities Commission of Ohio are offered in this tariff:

Standard Business Line Single and Multiple Analog PBX Trunk Digital Voice Grade DS-1 Trunk Service **Directory** Assistance **Operator Service** Local Calling Service Message Telecommunications Service **Custom Calling Features Trunk Side Features** Non Published Service 900/976 Blocking/Unblocking Maintenance Visit Charges **Directory** Listings Calling Card Services **Emergency Services Calling** Integrated Services Digital Network (ISDN) Primary Rate Interface (PRI) Direct Inward Dial (DID) Internet Service Provider (ISP) Standard Business Line Plus Digital Access Service (DAS) **Business Line Connect Dedicated Long Distance Services**

All services offered in this tariff are subject to order processing charges where the Customer requests new services or changes in existing services, as well as indicated Non-Recurring and Monthly Recurring Charges. Charges for Local Calling Service and Measured Telecommunications Service are assessed on a measured rate basis and are additional to the charges shown for Standard Residence/Business Line, Basic PBX Trunk and Digital Voice Grade DS-1 Trunk Service, as are other service charges.

Issued: June 18, 2001

Effective: July 18, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-1433-TP-ATA

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 (D)

5.2 <u>Standard Residence Line</u>

The Standard Residence Line provides a Customer with a single, analog, voice-grade telephonic communications channel which can be used to place or receive one call at a time. Standard Residence Lines are provided for the connection of Customer-provided wiring and single station sets or facsimile machines. *(See Section 6.2 for Price Ranges, See Section 10.1 for Price List Rate).*

5.3 <u>Standard Business Line</u>

The Standard Business Line provides a Customer with a single, analog, voice-grade telephonic communications channel which can be used to place or receive one call at a time. Standard Business Lines are provided for the connection of Customer-provided wiring and single station sets or facsimile machines. *(See Section 6.3 for Price Ranges See Section 10.2 for Price List Rate).*

5.4 Single and Multiple Analog PBX Trunk

Basic PBX Trunk Service provides a Customer with a single, voice-grade telephonic communications channel which can be used to place or receive one call at a time. Basic Trunks are provided for connection of Customer-provided private branch exchanges (PBX) to the public switched telecommunications network. Each Basic PBX Trunk is provided with touch tone signaling and may be configured into a hunt group with other Company-provided Basic PBX Trunks. The signal is an analog signal at the DSO level.

Basic Trunks may be equipped with Direct Inward Dialing (DID) capability and DID number blocks for additional charges. (See Section 6.4 for Price Ranges, See Section 10.3 for Price List Rate).

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

5.5 Digital Voice Grade DS-1 Trunk Service

Digital Voice Grade DS-1 Trunk Service provides business Customers with PBX or PBX-like equipment or other telephonic equipment with access to switch port. Customers can purchase this capability for both primary service (listed directory number) and diversity purposes. Customers must have the ability to terminate DS1 signal. The signal is delivered as a digital signal at the DS1 level.

The connection to the Customer's equipment is accomplished using a DS1 for digital connectivity.

Customers can subscribe to PBX Trunk Service - Digital/DS1 for local telecommunications services. Customers can also use this service for intraLATA and interLATA toll calling capability and for access to long distance carriers.

DID trunk signaling enables a PBX to switch an incoming call directly to the intended extension number without the need for an attendant. Digital Voice Grade DS-1 Trunk Service includes access to 911 service. (See Section 6.6 for Price Ranges, See Section 10.5 for Price List Rates).

5.6 Directory Assistance

Provides for identification of telephone directory numbers, via an operator or automated platform. Customers are provided with a maximum of 2 listings per each call to Directory Assistance. *(See Section 6.7 Price Ranges, See Section 10.6 for Price List Rates).*

5.7 **Operator Service**

[RESERVED FOR FUTURE]

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

5.8 Local Calling Service

Flat Service - A flat rate is in effect from originating NXX to terminating NXX.

Measured Service - Calls are billed in one (1) minute increments with an initial billing period of one (1) minute. Fractional minutes are rounded to the next full minute. (See Sections 6.2, 6.3, 6.4, and 6.6 for Price Ranges, See Sections 10.1, 10.2, 10.3, and 10.5 for Price List Rates)

5.9 Message Telecommunications Service

Message Telecommunications Service (MTS) is a communications service which is available for use by Customers twenty-four (24) hours a day. MTS enables a User of an exchange access line to place/receive calls to/from any station on the public switched telecommunications network bearing an NPA-NXX designation associated with points outside the Customer's Local Calling Area, but within the boundaries of the State of Ohio.

5.9.1 Switched Outbound Service

ICG Switched Outbound Service provides the customer with 1+ direct dial dialing. Except as otherwise provided below, ICG offers Switched Outbound Service to the following customer types:

- I. <u>Commercial Customers</u>
 - A. On-Switch
 - B. Resale
 - C. Stand-alone MTS Service offered to customers of ICG that do not have local Service with ICG.
- II. <u>Internet Enrollment Customers</u> MTS Service offered customers enrolled via the Internet.

5.9.1.1 IntraLATA Switched Outbound Service

IntraLATA Switched Outbound Service enables a User of an exchange access line to place calls to any station on the public switched telecommunications network bearing an NPA-NXX designation associated with points outside the Customer's Local Area, but within the LATA. *(See Section 6.9.1.1 for current pricing)*

5.9.1.2 InterLATA/Intrastate Switched Outbound Service

InterLATA/Intrastate Switched Outbound Service enables a User of an exchange access line to place calls to any station on the public switched telecommunications network bearing an NPA-NXX designation associated with points outside the Customer's Local Area, within the state, but outside the LATA. InterLATA/Intrastate Switched Outbound Service is an add-on to Interstate Switched Outbound Service. (See section 6.9.1.2 for current pricing)

Issued: June 18, 2001

Effective: July 18, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-1433-TP-ATA

5.9 Message Telecommunications Service (cont'd)

5.9.2 Switched Inbound Service (ICG Toll Free)

ICG Toll Free Service is an Inbound service which permits calls to a customer's station in one location from stations located elsewhere and in which the ICG Toll Free customer is billed for the calls rather than the call originator's. ICG offers Toll Free Service to the following customer types:

I. Commercial

II. Internet Enrollment

(D) (T)

5.9.2.1 IntraLATA Switched Inbound Service

IntraLATA Switched Inbound Service enables a User of an exchange access line to receive calls from any station on the public switched telecommunications network bearing an NPA-NXX designation associated with points within the LATA. IntraLATA Switched Inbound Service is an add-on to Interstate Switched Inbound Service. (See section 6.9.2.1 for current pricing)

5.9.2.2 InterLATA/Intrastate Switched Inbound Service

InterLATA/Intrastate Switched Inbound Service enables a User of an exchange access line to receive calls from any station on the public switched telecommunications network bearing an NPA-NXX designation associated with points outside the Customer's Local Area, within the state, but outside the LATA. InterLATA/Intrastate Switched Inbound Service is an add-on to Interstate Switched Inbound Service. *(See section 6.9.2.2 for current pricing)*

5.9.2.3 Per-Call Payphone Service Charge

Customer shall pay ICG a per-call service charge for all originating payphone traffic on ICG Toll Free Service. (See section 6.9.2.2 for current pricing)

5.10 <u>Custom Calling Features</u>

Call Waiting: permits the end-user engaged in a call to receive a tone signal indicating a second call is waiting; and, by operation of the switch hook, to place the first call on hold and answer the waiting call. The customer may alternate between the two calls by operation of the switch hook, but a three-way conference call cannot be established. *(See Section 6.10 for Price Ranges, See Section 10.9 for Price List Rates).*

Issued: June 18, 2001

Effective: July 18, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-1433-TP-ATA

5.10 Custom Calling Features (cont'd)

Three Way Calling: permits the end-user to add a third party to an established connection. When the third party answers, a two-way conversation can be held before adding the original party for a three-way conference. The end-user initiating the conference controls the call and may disconnect the third party to reestablish the original connection or establish a connection to a different third party. The feature may be used on both outgoing and incoming. (See Section 6.10 for Price Ranges, See Section 10.9 for Price List Rates).

Call Forwarding: permits the end-user to automatically forward (transfer) all incoming calls to another telephone number, and to restore it to normal operation at their discretion. . (See Section 6.10 for Price Ranges, See Section 10.9 for Price List Rates).

Call Forwarding Busy: permits the forwarding of incoming calls when the end-user's line is busy. The forwarded number is fixed by the end-user service order. *(See Section 6.10 for Price Ranges, See Section 10.9 for Price List Rates).*

Call Forwarding No Answer: permits the forwarding of incoming calls when the end-user's line remains unanswered after end-user-designated number of rings. The number of rings and the forwarded number are fixed by the customer service order. *(See Section 6.10 for Price Ranges, See Section 10.9 for Price List Rates).*

Speed Calling (8 or 30): permits the customer to place calls to other telephone numbers by dialing a one or two digit code rather than the complete telephone number. The feature is available as an eight code list or a thirty code list. Either code list may include local and/or toll telephone numbers. To establish or change a telephone number in a code list, the customer dials an activating code, receives a second dial tone and dials either a one or two digit code (for the eight code and thirty code lists, respectively) plus the telephone number. *(See Section 6.10 for Price Ranges, See Section 10.9 for Price List Rates).*

Vanity Number Service: Vanity Number Service is an optional feature by which a new Customer may request a specific or unique telephone number and fax number for use with the Company provided Exchange Services. This service provides for the assignment of a customer requested telephone number other than the next available number from the assignment control list.

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

5.10 Custom Calling Features (cont'd)

Vanity Number Service (cont'd)

Vanity Number Service is furnished subject to the availability of facilities and the requirements of Exchange Service as defined by the Company. The Company reserves all rights to the Vanity Numbers assigned to customer's and may, therefore, change them if required.

Monthly recurring charges apply per Vanity number. (See Section 6.10 for Price Ranges, See Section 10.9 for Price List Rates)

(M) (M)

Caller ID (w/o name): Caller ID allows a Customer to identify the telephone number from which a call is being made. The telephone number of the person initiating the call is displayed on a customer-provided display device unless the caller utilizes per call blocking or subscribes to per line blocking.

Caller ID (with name): Displays the name and telephone number of an incoming call. Information is displayed on a specialized CPE not provided by the Company.

Multi-Ring: Multi-Ring Service will be provided only in Telephone Company central offices where facilities permit.

Multi-Ring Service will enable a Customer to have as many as two telephone numbers associated with a single exchange service. Customers subscribing to this service will be able to receive calls dialed to two separate telephone numbers without having a second exchange service. A distinctive ringing pattern will be provided for each of the additional telephone numbers to facilitate identification of incoming calls.

(D)(D)

*Material Previously on this page has been moved to page 68.

Issued: February 19, 2001

Effective: March 1, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-249-TP-ATA

5.10 Custom Calling Features (cont'd)

911 Location Manager - 911 Location Manager (9LM) is an additional capability of E911 emergency response systems that provides end-users using a private telephone switch, such as a PBX or Centrex, with the Automatic Line Identification (ALI) feature(s) for individual telephone stations served by the private switch. This additional capability allows the private switch to identify individual subscriber address and location information to the Public Safety Answering Point (PSAP) operator. The customer maintains and updates their own ALI database information via the use of the third party vendor's website.

The 9LM feature capability consists of ALI database updates (i.e., the ability of the PBX end-user to update the ALI database) and the transport of 9LM calls to an E911 SR – Selective Router.

Customers must order the database functionality from third party vendor. Any customer premise equipment required to support this service must be provided by the customer. This feature is only available in conjunction with Fully Configured Digital Trunks and Digital Access Plan B service.

Issued: July 12, 2001

Effective: August12, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-1433-TP-ATA

(C)

5.0 NETWORK SERVICES DESCRIPTIONS - RETAIL PRICE LIST (cont'd)

5.10 Custom Calling Features (cont'd)

Per Call Blocking: Enable Customers to prevent the disclosure of their telephone number on per call basis to the called party. The disclosure of the calling party's number can be prevented on a per call basis by dialing a preassigned access code before making a call. This action must be repeated each time a call is made to prevent the disclosure of the calling party's telephone number. If the called party has a display device, a privacy indication will appear instead of the calling party's telephone number. Per Call Blocking will be provided on a universal basis to all eligible Customers.

Per Line Blocking: Enables Customers to prevent the disclosure of their telephone number on all outgoing calls, without the necessity of an activation code. If the called party has a display device, a privacy indication will appear instead of the calling party's telephone number. Per Line Blocking will be provided at no monthly charge on an optional basis to non-published Customers. The service is also available for subscription to published customers at a monthly charge of \$1.50 for Business Customers. Law Enforcement, Domestic Shelters and other special agencies will be offered free Per Line Blocking. Per Line Blocking will not be available to public, semi-public, and two-party service Customers.

Retained Number (DNCF) with 4 Paths: DNCF- Directory Number Call forwarding allows a customer to retain their current number. It will call forward to new telephone numbers issued by ICG. DNCF includes the capability to accept 4 call paths (concurrent calls) per retained number.

5.11 Trunk Side Features

Hunting: Routes a call to an idle station line in a prearranged group when the called station line is busy. (See Section 6.10 for Price Ranges, See Section 10.9 for Price List Rates)

Issued: June 18, 2001

Effective: July 18, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-1433-TP-ATA

5.0 NETWORK SERVICES DESCRIPTIONS (cont'd)

5.12 Non Published Service

This service provides for suppression of printed and recorded directory listings. (See Section 6.13 for Price Ranges, See Section 10.12 for Price List Rates).

5.13 900/976 Blocking/Unblocking

This service provides the option of blocking, or subsequent unblocking, all 900 and 976 calls on a per call and per line basis. The Company will provide for per-line blocking where the Company's switching facilities permit. (See Section 6.11 for Price Ranges, See Section 10.10 for Price List Rates).

5.14 Miscellaneous

5.14.1 Maintenance Visit Charges

Maintenance Visit Charges apply when the Company dispatches personnel to a Customer's premises to perform work necessary for resolving troubles reported by the Customer and the trouble is found to be caused by the Customer's facilities.

Maintenance Visit Charges will be credited to the Customer's account in the event trouble is not found in the Company facilities, but the trouble is later determined to be in those facilities.

The time period for which the Maintenance Visit Charges is applied will commence when Company personnel are dispatched at the Customer premises and end when work is completed. The rates for Maintenance of Service vary by time per Customer request. (See Section 6.12 for Price Ranges, See Section 10.11 for Price List Rates).

5.14.2 Order Processing Charge

Non-recurring charges apply to processing service orders for new service, for changes in service, and for changes in the Customer's PIC code. (See Section 6.12.2 for Price Ranges, See Section 10.11.2 for Price List Rates)

5.15 Directory Listings

For each Customer of Company-provided Exchange Service(s), the Company shall arrange for the listing of the Customer's main billing telephone number in the directory(ies) published by the dominant Local Exchange Carrier in the area at no additional charge. At a Customer's option, the Company will arrange for additional Business listings for an additional charge. *(See Section 6.13 for Price Ranges, See Section 10.12 for Price List Rates).*

*Material on this page was previously located on page 66.

Issued: February 19, 2001

Effective: March 1, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-249-TP-ATA

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 (T)

(T)

(M)

(M)

5.0 NETWORK SERVICES DESCRIPTIONS (cont'd)

5.16 Calling Card Services

The Company's Calling Card Services are offered to Customers of its local exchange (dial tone) services, such as Standard Business Line and Digital Voice Grade DS-1 Trunk Service. The Company's Calling Card Services are accessed by dialing a toll-free number and entering the Calling Card Number (are code, phone number, and PIN). *(See Section 6.14 for Price Ranges, See Section 10.13 for Price List Rates).*

5.17 Emergency Services Calling Plan

Access (at no additional charge by the Company) to emergency services by dialing 0 - or 9-1-1.

5.18 Integrated Services Digital Network (ISDN) Primary Rate Interface (PRI)

Integrated Services Digital Network (ISDN) is a set of transmission protocols that provides end-to-end digital connectivity and integration of voice, data and video, on a single Customer loop to support a wide variety of services via the public switched network. The Primary Rate Interface (PRI) consists of a 23B+D configuration with 23 64Kbps Bearer (B) digital channels and 1 64Kbps Data (D) digital channel. The B channels are designed for voice, data, image and sound transmissions. B channels can support synchronous, asynchronous or isochronous services at rates up to 64Kbps. B channels can be aggregated for higher bandwidth applications. The D channel PRI provides the out of band signaling, call control and messaging.

PRI is provided through standard four wire DS-1 (1.544Mbps) point to point, private line facilities that enables Customer control of the 24 individual channels. PRI supports 1+, 0+, 7 digit, and 10 digit IntraLATA and InterLATA services, as well as 01+ and 011+ international calling. PRI allows Customers to direct InterLATA voice, data and video over the Public Switched Telephone Network to the pre-subscribed IXC carrier of their choice, as well as 10XXX casual dialing. PRI also allows access to Public Switched Network services, such as Two-Way, Incoming Only, Outgoing Only and DID.\

Issued: June 18, 2001

Effective: July 18, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-1433-TP-ATA

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 (C)

(N)

(N)

5.0 NETWORK SERVICES DESCRIPTIONS - RETAIL PRICE LIST (cont'd)

5.18 Integrated Services Digital Network (ISDN) Primary Rate Interface (PRI) (cont'd)

Multiple PRI interfaces can be combined to function as one group. Utilizing a Backup D Channel arrangement, Customers are able to link up to 20 DS-1s together, providing a maximum of 479 64Kbps B channels controlled by a single D (signaling) channel.

In order to accommodate unusual network traffic loads created by the use of this service, the Company will not offer any new or additional PRI service, under the rates, terms, and conditions in effect prior to the effective date of October 1, 1998, to any Customer after the effective date of October 1, 1998. To further insure proper network capacity and assist Customers that desire additional capacity, the Company will waive the application of non-recurring charges for Customers that, as of the filing hereof, are already customers of the Company's PRI service and elect to migrate to the Company's Digital Access Service.

(See Section 6.6.3 for Price Ranges, See Section 10.5 for Price List Rates).

- **5.19** Direct Inward Dial (or "DID"): A service attribute that routes incoming calls directly to stations, by-passing a central answering point. (See Section 6.5 for Price Ranges, See Section 10.4 for Price List Rates).
- 5.20 <u>Internet Service Provider (ISP)</u>: A service that the Company offers to Customers wanting to access the Internet.
- 5.21 <u>Standard Business Line Plus</u>: Standard Business Line Plus is a local exchange service provided by ICG using Lucent Technologies 5ESS switches. It is classified as a central office switched service similar to a standard business line with the same identical calling patterns. The advantage to Standard Business Line Plus over a standard business line is the enhanced features it provides. Standard Business Line Plus offers the following features at no extra charge:

- DID to DOD Call Transfer	- Hunting	
----------------------------	-----------	--

- Three Way Calling
- Consultation Hold
- Touch Tone
- Consultation Hold

For an additional monthly charge Customers can order these additional features:

- Call Forward Variable Call Forward Busy
- Call Forward Don't Answer Message Waiting (sutter dial tone)

Standard Business Line Plus can be used with PBXs, Hybrids and Key Systems. It is an ideal product for new businesses and small to medium sized businesses. *(See Section 6.3.3 for Price Ranges, See Section 10.2.3 for Price List Rates).*

Issued: September 1, 1998

Effective: October 1, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

(N)

5.0 NETWORK SERVICES DESCRIPTIONS - RETAIL PRICE LIST (cont'd)

5.22 Digital Access Service

Digital Access Service (DAS) provides Customers with inbound and outbound digital access and transport capability to and from the public switched telephone network or other telecommunications facilities via a DS-1 signal. This Service is available under two distinct pricing plans, Plan A or Plan B, with the appropriate plan determined by the Company based on the Customer's usage profile as described herein.

This service is only available to Customers located in buildings that are served by ICG's network or in hybrid digital access buildings. This service is only available within calling areas served by ICG available NXXs. ICG reserves the right to reconfigure the network as necessary, which could include moving rate centers from one switch serving area to another, which may affect Customer charges.

Rates and charges for this service are exclusive of all other rates and charges for Company services. End users may incur local usage, toll or other charges in accordance with services they have purchased from the Company or their local exchange service provider.

This service is only available where facilities and operating conditions permit.

5.22.1 Plan A -- Inbound Services

This Service is for Customers whose traffic would otherwise pose a risk of call blocking on the Company's network due to unusually long call holding times and/or unusual traffic patterns. Where the Customer's usage of any Service offered by the Company is likely to be 80% or more inbound traffic and/or has an average per call holding time in excess of 30 minutes, the Customer's usage will be deemed within a potential blocking profile. The Company reserves the right to require any Customer who seeks to establish any new Service with the Company after the effective date of October 1, 1998 and whose traffic is forecast to be within this profile to purchase this Service if, in the judgement of the Company, the Customer's usage poses a significant risk of interfering with Service to other Customers of the Company by creating call blocking conditions.

(N)

Issued: September 1, 1998

Effective: October 1, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

NETWORK SERVICES DESCRIPTIONS (cont'd) 5.0

5.20 Digital Access Service (cont'd)

5.20.1 Plan A — Inbound Services (cont'd)

Plan A service arrangements are designed for data applications, which terminate into Customer Premises Equipment (CPE).

This Service does not provide for E-911 or Operator Services connectivity and the Customer must separately arrange for Local Exchange Service from the Company or another certificated local exchange service provider to obtain such connectivity.

5.20.2 Plan B

Plan B — Inbound, Outbound, Two-way Services 5.20.2.1

This Service is for Customers whose traffic patterns are other than those described in 5.20.1. Plan B can be arranged as inward-only, including Direct Inward Dialing service, outward-only, or Twoway, and may include voice or data applications.

Plan B — High Volume 5.20.2.2

This service is for customers anticipating high volume outward-only or two way traffic. A minimum of 20 Plan B High Volume PRI's is required per customer, the minimum term for this option is 1 year. A per minute average rate will apply should customers exceed allotted minutes of use. Customers must purchase a Digital Access Loop at the current tariffed rate. Caller ID and Hunting will be included, all other features may be added at the current tariffed rate. EOS is not available with this option.

5.20.3 Plan C --- National Services

This service is for Plan A Customers whose aggregate volume meets or exceeds 200 PRI's in multi-state locations. The Customer must contract for a minimum two year term in order to receive the volume discounts.

A Rate Stability option is available for Plan A Customers whose aggregate volume meets or exceeds 250 PRI's. The customer must contract for a minimum three year term in order to be eligible for the Rate Stability plan.

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112

Effective: September 24, 2001

(T)

(T)



(N)

5.0 NETWORK SERVICES DESCRIPTIONS (cont'd)

5.20 Digital Access Service (cont'd)

5.20.4 Plan D --- EOS/DS-3 Component

This pricing component provides a circuit equivalent to 28 DS-1's and operates at 44.736 mbps. This pricing component replaces the Digital Access Loop, EOS IntraLATA transport and/or EOS InterLATA transport provisioned at the DS-1 level and available in requested markets

5.20.4.1 Plan E – Regional Services

This service is for Plan A Customers whose aggregate volume falls into the following categories; 25-49 PRI, 50-99 PRI or 100+ PRI, in one or multi-state locations.

5.20.4.2 Plan F - Long-Term Collocated Service

To obtain this service a Plan A customer must have collocation in every ICG office and provide service in all ICG markets. The customer will be obligated for an minimum term of five years and must provide aggregated volumes from 200,000 to 500,000 PRI within multi state locations. This plan is only available for agreements signed prior to January 4, 2002.

(N)

5.20.4.3 Plan G- Long-Term, High Volume, Multi-state Service

To obtain this service a Plan A customer must maintain collocation in every ICG office, where feasible, and provide service in all ICG markets. The customer will be obligated for a minimum 66 month term and must provide aggregated volumes of at least 100,000 PRI.

If a Plan G customer chooses to terminate PRI circuits prior to the end of the term of the agreement, the following termination liability will apply :

During the first year of the agreement: 100% of the MRC for the canceled PRI circuits multiplied by the number of months remaining in the term.

During the second and third years: 85% of the MRC for the canceled PRI circuits multiplied by the number of months remaining in the term.

During the fourth and fifth years: 70% of the MRC for the canceled PRI circuits mulitplied by the number of months remaining in the term.

* Material previously found on this page is now found on page 70.3.1.

Issued: December 4, 2001

Effective: January 4, 2002

(M)

(M)

5.0 <u>NETWORK SERVICES DESCRIPTIONS</u> (cont'd)

5.20.5 Rate Element Descriptions --- Standard Features (Plans A and B)

A. DS1 Central Office Access Port. The port is used to connect the Digital Access Loop to the ICG serving switch. Monthly recurring and non-recurring installations charges apply per DS1 ordered. Three access port options are available.

* Material found on this page was previously on page 70.3.

Issued: December 4, 2001

Effective: January 4, 2002

5.24 Dedicated Long Distance Service

Dedicated Long Distance consists of circuits "dedicated" to long distance traffic connecting the customer's premises with ICG's dedicated long distance points of presence. Dedicated long distance will only be offered to customers within the ICG local services footprint, in both the Ameritech and Cincinnati Bell Serving areas. Dedicated long distance does not utilize a local exchange carrier's switch. Dedicated long distance provides only voice grade circuits.

Fractional cents will be rounded to the nearest cent. For example, if the customer's call is less than \$0.085, the charge is \$0.08. If the customer's call is equal to or higher than \$0.085, the charge is \$0.09.

5.24.1 Plan A

This plan includes Dedicated Inbound and Dedicated Outbound Services. Customers subscribing to this plan will receive a 40% discount on all IntraLATA/Intrastate and InterLATA/Intrastate calls. See Section 6.18 for price ranges, see section 10.17 for Price List Ranges.

5.24.2 Plan B

TBD

5.25 Business Essentials

Business Essentials provides the customer with a bundled solution for their local and long distance service. The Business Essentials bundle contains one Standard Business Line Plus line including the following features; hunting, 3 Way calling, call transfer, consultation hold, Caller ID, Call Waiting, and Call Forwarding Variable/Busy/ No Answer. Business Essentials also includes a discounted long distance rate for all long distance calls (IntraLATA/intrastate, InterLATA/intrastate, interstate, outbound and inbound calls), calling card and international calls are not included and are charged at the regular tariffed rates.

Business Essentials rates and services are only available with the bundled offering. Early termination penalties apply for cancellation of any component of the bundle prior to the end of the term of the agreement. Customers may upgrade their service at any time without penalty.

In Compliance with Rule 4901:1-5-16(B), O.A.C., the monthly rate for a Standard Business Line (N) in the Ameritech serving area is \$19.90; the monthly rate for a Standard Business Line in the Cincinnati Bell Serving area is \$38.40. (N)

Issued: March 21, 2000

Effective: March 22, 2000

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF

6.0 NETWORK SERVICES - RETAIL PRICE LIST

6.1 <u>Service Offerings</u>

The Retail Rate Schedule for Network Services provides for local dialtone service and related switched services.

The following Network Services are offered in this tariff:

Standard Residence Line Standard Business Line Single and Multiple Analog PBX Trunk Direct Inward Dial (DID) Digital Voice Grade DS-1 Trunk Service Integrated Services Digital Network (ISDN) Primary Rate Interface (PRI) **Directory** Assistance **Operator Service** Message Telecommunications Service **Custom Calling Features** 900/976 Blocking/Unblocking Maintenance Visit Charges **Directory** Listings Calling Card Services Standard Business Line Plus Digital Access Service (DAS) **Business Line Connect** Dedicated Long Distance Services

All services offered in this tariff are subject to order processing charges where the Customer requests new services or changes in existing services, as well as indicated Non-Recurring and Monthly Recurring Charges. Charges for Local Calling Service and Measured Telecommunications Service are assessed on a measured rate basis and are additional to the charges shown for Standard Residence/Business Line, Single and Multiple Analog PBX Trunk, Digital Voice Grade DS-1 Trunk Service, as are other service charges.

Issued: February 9, 2001

Effective: March 1, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-249-TP-ATA

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 (D)

- 6.3 Standard Business Line (see Section 10.2 of this tariff for Price List Rates).
 - 6.3.1 Single Line and Multiple Line Business Customers (see Section 10.2.1 of this tariff for Price List Rates).

Monthly Recurring Line Charge, Ameritech Serving Area (Touch Tone Included):

Term Plan	Minimum	Maximum
Month to Month	\$16.92	\$ 20.90
12 Month	\$16.58	\$20.48
24 Month	\$16.36	\$20.21
36 Month	\$15.98	\$19.74
48 Month*		
60 Month*		

(D)

(D)

Monthly Recurring Line Charge, Cincinnati Bell Service Area (Touch Tone Included):

Term Plan	Minimum	Maximum
Month to Month	\$19.20	\$ 76.80
12 Month	\$17.35	\$ 69.40
24 Month	\$16.65	\$ 66.60
36 Month	\$16.20	\$ 64.80

(D)

(N)

(N)

(D)

*48 and 60 month customers with service contracts prior to April 1, 1999, will not experience changes in their service rates or conditions for the duration of their contract. Term Plans reflected above, at current rates and conditions will occur if the customer chooses to renew their contract after April 1, 1999.

Issued: March 1, 1999

Effective: April 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-1650-TP-TRF.

161 In Engley	elecom Group, Inc. verness Drive West wood, CO 80112			PUCO Tariff No. 2 nd Revised Page 7 els 1 st Revised Page 7	8
6.0 6.3	NETWORK SERVICES - RETAIL Standard Business Line (cont'd) (see			tes).	
	6.3.1 Single Line and Multiple lon	e Business Custome	e <u>rs</u> (see Section Price I	10.2.1 of this tariff for List Rates).	(T)
	Other Standard Business Li	ne Charges and Cre <u>Mo</u>	dits: nthly Recurrin	g Charges	(M)
	911 Emergency Charge	Minimu	m	Maximum	(D)
	Ameritech Serving Area Cincinnati Bell Serving Area *Does not apply to DID or inboun	\$0.10 \$0.12 d only equipped circu		\$0.126 \$0.48	(N)
	TDD Surcharge	\$0.00		\$0.00	(NI)
	Monthly Recurring Line Charge V	olume Discounts:	None		I
	Non Recurring Installation Charge	e (Ameritech Serving	g Area):		(T)
	Per Order - 1 st Line: Additional Lines		<u>Minimum</u> \$38.25 \$17.00	<u>Maximum</u> \$60.00 \$25.00	(M)
	Local Per Call Charges (Ameritech	1 Serving Area):	<u>Minimum</u> \$0.051	<u>Maximum</u> \$0.063	(T)
	Non Recurring Installation Charge	e (Cincinnati Bell Se	rving Area):*	*	(N)
	Per Order - 1 st Line: Additional Lines		<u>Minimum</u> \$1.00 \$1.00	<u>Maximum</u> \$90.00 \$40.00	 (N)
In See	traLATA IntraState Toll Charges: Section 6.9.1 for Price Ranges, See Section 10	9.8.1 for Price List Rates.			
	ICG Toll Rounding Policy:	18 seconds			

18 seconds Call Minimum Call Rounding 6 seconds * Local Calling Service Areas are listed in Section 4.1 of this tariff. **The Company may waive these charges by publishing a Tariff Price List reflecting no charge.

Issued: December 11, 1998

Effective: December 11, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

6.3 Standard Business Line (cont'd) (see Section 10.2 of this tariff for Price List Rates).

*Material previously located on this page is now located on Page 78.

Standard Business Line Early Termination Penalty Charge:

If the Customer terminates their Standard Business Line Term Plan prior to its completion, the Customer will be responsible for reimbursing ICG for the difference between the Term Plan monthly recurring charge and the Month-to-Month monthly recurring charge for all completed or partial months from the beginning of the agreement period for each of the Customers' Standard Business Lines covered under the Term Plan. In addition, the Customer must reimburse ICG for all installation, wiring or other initial fees that may have been waived at time of initial service.

Issued: December 11, 1998

Effective: December 11, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

- 6.3 <u>Standard Business Line</u> (cont'd) (see Section 10.2 of this tariff for Price List Rates).
 - 6.3.3 ICG Standard Business Line Plus (see Section 10.2.3 of this tariff for Price List Rates).

Monthly Recurring Line Charge, Ameritech Serving Area (Touch Tone Included):

Term Plan	Minimum	Maximum
Monthly	\$16.92	\$20.90
12 Month	\$16.58	\$20.48
24 Month	\$16.36	\$20.21
36 Month	\$15.98	\$19.74
48 Month*		
60 Month*		

Monthly Recurring Line Charge, Cincinnati Bell Serving Area (Touch Tone Included):

Term Plan	Minimum	Maximum
Monthly	\$20.35	\$81.40
12 Month	\$19.65	\$78.60
24 Month	\$18.00	\$72.00
36 Month	\$17.33	\$69.30

*48 and 60 month customers with service contracts prior to April 1, 1999, will not experience changes in their service rates or conditions for the duration of their contract. Term Plans reflected above, at current rates and conditions will occur if the customer chooses to renew their contract after April 1, 1999.

Issued: March 1, 1999

Effective: April 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-1650-TP-TRF.

- 6.3 Standard Business Line (cont'd) (see Section 10.2 of this tariff for Price List Rates).
 - 6.3.3 ICG Standard Business Line Plus (see Section 10.2.3 of this tariff for Price List Rates).

Other Standard Business Line Cha	arges and Credits: Monthly Recurring C	harges
	Minimum	Maximum
911 Emergency Charge		
Ameritech Serving Area Cincinnati Bell Serving Area *Does not apply to DID or inbound only	\$0.102 \$0.12 equipped circuits	\$0.126 \$0.48
TDD Surcharge Standard Business Line Plus Features Pa	\$0.00 ckage:	\$0.00
3-Way Calling Consultation Hold Touch Tone Call Transfer Hunting	N/C N/C N/C N/C N/C	

Business Line Optional Features, Ameritech Serving Area:¹

Monthly Recurring Charge

	Minimu <u>m</u>	<u>Maximum</u>
Call Forward Variable	\$1.28	\$1.58
Call Forward Busy	\$1.28	\$1.58
Call Forward No Answer	\$1.28	\$1.58
Message Waiting	\$1.28	\$1.58

¹ This feature package is available only to customers who have signed service contracts prior to 4/1/99. Please refer to Section 6.10 of this tariff for price ranges on Custom Calling Features for service contracts signed after 4/1/99.

*Local Calling Service Areas are listed in Section 4.1 of this tariff.

Issued: March 1, 1999

Effective: April 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-1650-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112 (C)

Single and Multiple Analog PBX Trunk (see Section 10.3 of this tariff for Price List Rates). 6.4

On-Switch PBX DS-0 Trunk Volume Discounts, Ameritech Serving Area: Volume discounts are based on the number of lines and term plan as described below.

Monthly Recurring PBX DS-0 Trunk Charges, Ameritech Serving Area (Includes Touch Tone):

_	(Includes rouch rone).											
	Mont Mot		12 M	[onth	24 M	onth	36 M	lonth	48 M	onth	60 M	Ionth
Trunks	Min	Max	Min	Max	Min	Max	Min	Max	Min	Max	Min	Max
1-47	\$21.56	\$26.34	\$21.37	\$26.40	\$21.04	\$25.99	\$20.55	\$25.39	\$20.26	\$25.03	\$19.93	\$24.62
48-96	\$20.96	\$25.89	\$20.77	\$25.66	\$20.45	\$25.26	\$19.92	\$24.60	\$19.70	\$24.34	<u>\$19.38</u>	\$23.94
97-199	\$20.37	\$25.16	\$20.19	\$24.94	\$19.87	\$24.55	\$19.35	\$23.90	\$19.14	\$23.65	\$18.83	\$23.26
200-293	\$19.76	\$24.41	\$19.59	\$24.20	\$19.30	\$23.84	\$18.79	\$23.21	\$18.58	\$22.95	\$18.28	\$22.58
294-387	\$19.17	\$23.68	\$19.01	\$23.48	\$18.71	\$23.11	\$18.22	\$22.50	\$18.02	\$22.26	\$17.72	\$21.89
388-579	\$18.56	\$22.93	\$18.41	\$22.74	\$18.13	\$22.40	\$17.65	\$21.81	\$17.46	\$21.57	\$17.18	\$21.22
580+	\$17.97	\$22.20	\$17.82	\$22,02	\$17.54	\$21.67	\$17.09	\$21.11	\$16.90	\$20. <u>8</u> 7	\$16.63	<u>\$20.54</u>

Applies to inbound, outbound, DID, DOD, 2-Way DID circuits.

Term Plan

Monthly Recurring PBX DS-0 Trunk Charges, Cincinnati Bell SerServing Area (Includes Touch Tone): Minimum Maximum

(N)

Month to Month12 Month24 Month36 Month48 Month60 Month	\$ 29.50 \$ 28.50 \$ 28.00 \$ 27.50 \$ 26.50 \$ 25.50	\$ 118.50 \$ 114.00 \$ 112.00 \$ 110.00 \$ 106.00 \$ 102.00		
Other Monthly Recurring Charges: <u>911 Emergency Charge</u> Ameritech Serving Area Cincinnati Serving Area *Does not apply to DID or inbound only equipped circuits End-User Common Line (EUCL)**		<u>Minimum</u> \$0.102 \$0.12 \$4.52	<u>Maximum</u> \$0.126 \$0.48 \$5.59	(N) (T) (D) (T) (N)
*Also known as FCC Charge TDD Surcharge **The Company may waive these charges by publishing a Tari	iff Price List refl	\$0.00 ecting no charge	\$0.00	

Issued: December 11, 1998

Effective: December 11, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112

(T)

(T)

161 In	elecom Group, Inc. verness Drive West vood, CO 80112	2 ^r Cancels	PUCO Tariff No. 2 nd Revised Page 82 1 st Revised Page 82	
6.0	NETWORK SERVICES - RETAIL PRICE LIST	<u>[</u> (cont'd)		
6.4	Single and Multiple Analog PBX Trunk (cont'd) (see Section 10.3 of this tariff for Price List Rates).			(T)
	Ameritech Serving Area			
	Monthly Recurring Line Charge Volume	Discounts: N	lone	
	Non Recurring Installation Charge:	Minimum	Maximum	
	Per Order - 1 st Line Additional Lines	\$38.25 \$17.00	\$60.00 \$25.00	
	Local Per Call Charges	<u>Minimum</u> \$0.051	<u>Maximum</u> \$0.063	
	<u>Cincinnati Bell Serving Area</u>			(N)
	Non Recurring Installation Charge**	Minimum	Maximum	
	Per Order	\$1.00	\$120.00	
Int	raLATA IntraState Toll Charges:			(N)

See Section 6.9.1 for Price Ranges, See Section 10.8.1 for Price List Rates.

ICG Toll Rounding Policy:

Call Minimum 18 seconds Call Rounding 6 seconds * Local Calling Service Areas are listed in Section 4.1 of this tariff.

PBX DS-0 Trunk Early Termination Penalty Charge:

If the Customer terminates their PBX DS-0 Trunk Term Plan prior to its completion, the Customer will be responsible for reimbursing ICG for the difference between the Term Plan monthly recurring charge and the Month-to-Month monthly recurring charge for all completed or partial months from the beginning of the agreement period for each of the Customers' PBX DS-0 Trunks covered under the Term Plan. In addition, the Customer must reimburse ICG for all installation, wiring or other initial fees that may have been waived at time of initial service.

**The Company may waive these charges by publishing a Tariff Price List reflecting no charge.

Issued: December 11, 1998

Effective: December 11, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

6.5 Direct Inward Dial Service (DID) (see Section 10.4 of this tariff for Price List Rates).

Monthly Recurring Charges, Ameritech Serving A Per DID Equipment Line/Trunk Charge	Area: <u>Minimum</u> \$17.00	<u>Maximum</u> \$21.00	
1 st Block of 20 DID Numbers	\$ 2.34	\$ 2.89	
Additional Blocks of 20 DID Numbers	\$ 2.34	\$ 2.89	
Monthly Recurring Charges, Cincinnati Bell Serv	ing Area:	Mar	
Per DID Trunk	<u>Min</u> \$ 5.00	<u>Max</u> \$20.00	
Per Block of 20 DID Numbers	\$ 1.38	\$ 5.50	
Additional block of 20 DID numbers	\$ 1.50	\$ 6.00	(N)
1 st Block of 20 DID numbers over 100	\$ 1.00	\$60.00	(N)
Non-Recurring DID Installation Charges, Amerit	ech Serving Are Minimum	a: Maxi <u>mum</u>	
Per DID Equipped Line/Trunk Charge	\$187.00	\$231.00	
1 st Block of 20 DID Numbers	\$ 85.00	\$105.00	
Additional Blocks of 20 DID Numbers	\$ 29.75	\$ 36.75	
Non-Recurring DID Installation Charges, Cincing	nati Bell Serving <u>Min</u>	g Area:** Max	
Per DID Trunk	\$ 1.00	\$ 20.00	
Per Block of 20 DID Numbers	\$ 1.00	\$ 200.00	
Additional Blocks of 20 Numbers	\$ 1.00	\$ 40.00	(N)
1 st Block of 20 Numbers over 100	\$ 1.00	\$ 40.00	(N)

*DID Equipped Line/Trunk MRCs are in addition to the Standard Business Line MRCs, the PBX DS-0 Trunk MRCs, or the Digital DS-1 Trunk active channel MRCs. In addition, the Customer will be charged for the number of DID Number Blocks (20 numbers per block) regardless of the number of DID numbers utilized out of the available 20 numbers.

**The Company may waive these charges by publishing a Tariff Price List reflecting no charge.

Issued: March 1, 1999

Effective: April 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-____-TP-TRF.

6.5 Direct Inward Dial Service (DID) (cont'd) (see Section 10.4 of this tariff for Price List Rates).

6.5.1 <u>Direct Outward Dial (DOD) and 2-Way DID Service.</u>			
Ameritech Serving Area	<u>Minimum</u>	(T) <u>Maximum</u>	
Monthly Recurring Charges: Per DOD Equipped Line Trunk Charge	\$0.00	\$0.00	
Per 2-Way DID Equipped Line/ Trunk Charge	\$17.00	\$21.00	
1 st Block of 20 DID Numbers	\$ 2.34	\$ 2.89	
Additional Blocks of 20 DID Numbers	\$ 2.34	\$ 2.89	
Non-Recurring DOD and 2-Way DID Installation Charges:			
• –	<u>Minimum</u>	<u>Maximum</u>	
Per DOD Equipped Line/ Trunk Charge	\$ 0.00	\$ 0.00	
Per 2-Way DID Equipped Line/ Trunk Charge	\$187.00	\$231.00	
1 st Block of 20 DID Numbers	\$ 85.00	\$105.00	
Additional Blocks of 20 DID Numbers	\$ 29.75	\$ 36.75	

*DOD and 2-Way DID Equipped Line/Trunk MRCs are in addition to the Standard Business Line MRCs, the PBX DS-0 Trunk MRCs, or the Digital DS-1 Trunk active channel MRCs. In addition, the Customer will be charged for the number of DID Number Blocks (20 numbers per block) regardless of the number of DID #s utilized out of the available DID 20 number blocks.

Issued: December 11, 1998

Effective: December 11, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

6.6 Digital Voice Grade DS-1 Trunk Service (see Section 10.5 of this tariff for Price List Rates).

Digital Voice Grade DS-1 Trunk Service provides line connections from an end user's Private Branch Exchange ("PBX"), Key System, or other telephonic device to the Switch Port. Digital Voice Grade DS-1 Trunk Service is available as a DS1 (1.544 Mbps). Service on a Digital Trunk is available on a measured rate basis for business customers only. A minimum order of 12 voice grade connections (active DSO trunks) is required. Charges for Local Calling Service and Message Telecommunications Service are assessed on a measured rate basis and are additional to charges for Digital Voice Grade DS-1 Trunk Service. These charges only apply to ICG Customers. *(see Sections 6.4 and 6.9 for Price Ranges, see Sections 10.3 and 10.8 for Price Lists).*

The Customer may opt to utilize Digital Voice Grade DS-1 Trunk Service for outgoing calls only, or for outgoing and incoming calls. In the latter case, direct inward dial ("DID") numbers must be purchased from the Company.

6.6.1.1 Digital Voice Grade DS-1 Trunk Service (24 Channels),

Ameritech Serving Area (see Section 10.5.1 of this tariff for Price List Rates).

(T)

Monthly Recurring DS-1 Trunk Charges Per DS-1 (24 Channel Ready) Circuit

	nth to onth	12 N	/Ionth	24 N	Ionth	36 N	lonth	48 N	1onth	60 N	Ionth
Min	Max	Min	Max	Min	Max	Min	Max	Min	Max	Min	Max
		<u> </u>			0160.00	075.00	\$160.00	\$75.00	\$160.00	\$75.00	\$160.00
\$75.00	\$160.00	\$75.00	\$160.00	\$75.00	\$160.00	\$75.00	\$100.00	915.00	<u></u>	W15.00	

Issued: December 11, 1998

Effective: December 11, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

6.6 Digital Voice Grade DS-1 Trunk Service (cont'd) (see Section 10.5 of this tariff for Price List Rates).

6.6.1.1 Digital Voice Grade DS-1 Trunk Service (24 Channels), <u>Ameritech Serving Area</u> (cont'd) (see Section 10.5.1 of this tariff for Price List Rates).

Monthly Recurring Active DS-1 Channel Trunk Charges:

<u>Term Plan</u>	<u>Minimum</u>	<u>Maximum</u>
Month to Month	\$18.21	\$27.68
12 Month	\$18.21	\$27.68
24 Month	\$18.21	\$27.68
36 Month	\$15.00	\$27.68 (R)
48 Month	\$14.00	\$27.68 (R)
60 Month	\$13.00	\$27.68 (R)

Applies to all active inbound, outbound, DID, DOD, 2-Way DID DS-1 channels Hunting Charges do not apply

Other Trunk Service Charges and Credits:

	Monthly Recurring Charges	
	<u>Minimum</u>	<u>Maximum</u>
<u>911 Emergency Charge</u> Ameritech Serving Area Cincinnati Serving Area *Does not apply to DID or inbound only equip	\$0.102 \$0.12 ped circuits	\$0.126 \$0.48
TDD Surcharge	\$0.00	\$0.00
Monthly Recurring DS-1 Trunk Volume Discounts:	None	
Monthly Recurring DS-1 Trunk Volume Discounts: Non Recurring Installation Charge: <u>Per Order</u>	None <u>Minimum</u>	<u>Maximum</u>

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

6.6 <u>Digital Voice Grade DS-1 Trunk Service</u> (cont'd) (see Section 10.5 of this tariff for Price List Rates).

6.6.1.1 <u>Digital Voice Grade DS-1 Trunk Service (24 Channels)</u> <u>Ameritech Serving Area</u> (cont'd) (see Section 10.5.1 of this tariff for Price List Rates).

	<u>Minimum</u>	<u>Maximum</u>
Local Per Call Charges	\$0.051	\$0.063

IntraLATA IntraState Toll Charges:

See Section 6.9.1 for Price Ranges, See Section 10.8.1 for Price List Rates.

ICG Toll Rounding Policy:	
Call Minimum	18 seconds
Call Rounding	6 seconds

* Local Calling Service Areas are listed in Section 4.1 of this tariff.

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 İ

(D)

(D)

6.6 <u>Digital Voice Grade DS-1 Trunk Service</u> (cont'd) (see Section 10.5 of this tariff for Price List Rates).

6.6.1.2 DS-1 Trunk Charges - Cincinnati Bell Service Area

A) Basic Digital Trunk Option

The basic Digital trunk option must include a DS-1 loop connection, unless customer is collocated with an ICG facility.

1) DS-1 Loop Connection

Monthly Recu	rring Charge	<u>Non-R</u>	ecurring Charge
Min	\$75.00	Min	\$250.00
Max	\$300.00	Max	\$1000.00

2) DS0 Charge - Applies to 2-way, DID or DOD

Non-Recurring Charge (Per Active DS0):

1 st DS	0 per order	Each a	additional DS0 per order
Min	\$22.50	Min	\$10.00
Max	\$90.00	Max	\$40.00

Monthly Recurring Charge (Per Active DS0):

	Min	Max
Month-to-Month	\$17.50	\$70.00
12 Month	\$17.00	\$68.00
24 Month	\$15.50	\$62.00
36 Month	\$14.50	\$58.00
48 Month	\$14.25	\$57.00
60 Month	\$13.50	\$54.00

B) Fully Configured Digital Trunk Option - Includes DS-1 Loop and 24 Active DS0s

Non-Recurring Charges				
Min	\$500.00			
Max	\$2000.00			

Monthly Recurring Charge (Per DS-1)

	Min	Max
Month-to-Month	\$489.00	\$1956.00
12 Month	\$472.00	\$1888.00
24 Month	\$383.00	\$1532.00
36 Month	\$348.00	\$1392.00
48 Month	\$331.00	\$1324.00
60 Month	\$314.00	\$1256.00
24 Month 36 Month 48 Month	\$383.00 \$348.00 \$331.00	\$1532.00 \$1392.00 \$1324.00

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

6.6 <u>Digital Voice Grade DS-1 Trunk Service</u> (cont'd) (see Section 10.5 of this tariff for Price List Rates).

6.6.1.3 Three Way Calling/Call Transfer Feature, Ameritech and Cincinnati Bell Serving Areas

Available on digital trunks only, this feature provides functionality allowing a station user to transfer any established call to another station outside the PBX or business group without the assistance of the attendant. The customer may also add on another party for a three way call. Customers utilizing this service will be billed for any toll charges applicable to the call.

Min.\$25.00 per trunk groupMin.\$1.50 per trunkMax.\$100.00 per trunk groupMax..\$6.00 per trunk

(D)

(D)

(D)

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

6.6 Digital Voice Grade DS-1 Trunk Service (cont'd)

(see Section 10.5 of this tariff for Price List Rates).

6.6.2.1 <u>Fully Configured Digital Trunk option - includes DS-1 Loop and 24 active DS0s</u> <u>Ameitech Serving Area</u>

Non Recurring ChargesMinMax\$200.00 \$1500.00

Monthly recurring charge (per DS-1)

	Min	Max
Month-to-Month	\$345.00	\$1380.00
12 Month	\$298.00	\$1192.00
24 Month	\$245.00	\$ 980.00
36 Month	\$210.00	\$ 840.00
48 Month	\$199.00	\$ 796.00
60 Month	\$194.00	\$ 776.00

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 (N) | | | | | |

(D)

(D)

(N)

(T)

(T)

(D)

(D)

6.0 NETWORK SERVICES - RETAIL PRICE LIST (cont'd)

6.6 <u>Digital Voice Grade DS-1 Trunk Service</u> (cont'd)

(see Section 10.5 of this tariff for Price List Rates).

6.6.2.1 <u>ISP Monthly Recurring Active DS0 charge - applies to 2-way, DID, DOD, Ameritech Serving</u> <u>Area</u>

<u>Term Plan</u>	Minimum	<u>Maximum</u>
Month to Month	\$12.08	\$16.77
12 Month	\$12.08	\$16.77
24 Month	\$12.08	\$16.77
36 Month	\$12.08	\$16.77
48 Month	\$12.08	\$16.77
60 Month	\$12.08	\$16.77

*Hunting Charges do not apply

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

6.6 Digital Voice Grade DS-1 Trunk Service (cont'd) (see Section 10.5 of this tariff for Price List Rates).

6.6.2.1 <u>ISP Monthly Recurring Active DS0 charge - applies to 2-way, DID, DOD, Ameritech Serving</u> (N) (N) (N) (D)

Local Per Call Charges	

<u>Minimum</u> \$0.051 <u>Maximum</u> \$0.063 (D)

IntraLATA IntraState Toll Charges:

See Section 6.9.1 for Price Ranges, See Section 10.8.1 for Price List Rates.

ICG Toll Rounding Policy:

Call Minimum Call Rounding 18 seconds 6 seconds

* Local Calling Service Areas are listed in Section 4.1 of this Tariff.

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

(T)

6.0 NETWORK SERVICES - RETAIL PRICE LIST (cont'd)

6.6 <u>Digital Voice Grade DS-1 Trunk Service</u> (cont'd) (see Section 10.5 of this tariff for Price List Rates).

6.6.2.3 Digital DS-1 Trunk and Active Channel Early Termination Penalty Charge:

If the Customer terminates their Digital Data DS-1 Trunk and Active Channel Term Plan prior to its completion, the Customer will be responsible for reimbursing ICG for the difference between the Term Plan monthly recurring charge and the Month-to-Month monthly recurring charge for all completed or partial months from the beginning of the agreement period for each of the Digital Data DS-1 Trunk and Active Channel covered under the Term Plan. In addition, the Customer must reimburse ICG for all installation, wiring or other initial fees that may have been waived at time of initial service.

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

6.6 <u>Digital Voice Grade DS-1 Trunk Service</u> (cont'd) (see Section 10.5 of this tariff for Price List Rates).

6.6.3 <u>Integrated Services Digital Network (ISDN) Primary Rate Interface (PRI)</u> (see Section 10.5.3 of this tariff for Price List Rates).

Monthly Recurring Charge:

	Moi	nthly	12 M	onth	24 M	lonth	36 M	lonth	48 M	lonth	60 M	lonth
Trunks	Min	Max										
1 st PRI	\$297.50	\$367.50	\$289.00	\$357.00	\$280.50	\$346.50	\$272.00	\$336.00	\$267.75	\$330.75	\$263.50	\$325.50
2 nd PRI	\$212.50	\$262.50	\$204.00	\$252.00	\$195.50	\$241.50	\$187.00	\$231.00	\$182.75	\$225.75	\$178.50	\$220.50
3 rd PRI	\$170.00	\$210.00	\$16150	\$199.50	\$153.00	\$189.00	\$144.50	\$178.50	\$140.25	\$173.25	\$136.00	\$163.00

• ISDN PRI MRC's are in addition to the Digital T1 Facility MRC's (See Section 6.6.1 of this tariff for Digital T1 Monthly Recurring Charges).

ISDN PRI installation Charge:	Minimum	<u>Maximum</u>
Per T1 Facility	\$1,487.50	\$1,83
·		7.50

Term Plan Price Standard Features Include:

- 23B+D
- 24B; Requires purchase of one 23B+D
- "D" Channel Control of Multiple ISDN Primes
- 64 Clear Channel Capability
- Direct Inward Dialing
- Dedicated Trunk Groups

DS-1 Trunk charge is additional; Refer to Local Service pricing for DS-1 Circuit

DID Trunk Termination:

DID channel termination is eliminated do to the non-channelization of the PRI trunk. A monthly fee for each telephone number will apply - refer to the DID pricing portion of this tariff. (See Section 6.5 for Price Ranges, See Section 10.4 for Price List Rates).

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112

6.6 <u>Digital Voice Grade DS-1 Trunk Service</u> (cont'd) (see Section 10.5 of this tariff for Price List Rates).

6.6.3 <u>Integrated Services Digital Network (ISDN) Primary Rate Interface (PRI)</u> (cont'd) (see Section 10.5.3 of this tariff for Price List Rates).

ISDN PRI Optional Feature: Monthly Recurring Charge:

23B with "D" Channel Backup

<u>Minimum</u> \$85.00 <u>Maximum</u> \$115.50

(The 23 B is provisioned separately as a 23B+D; The D Channel Backup is provisioned separately)

	Installation		
	<u>Minimum</u>	<u>Maximum</u>	
Additions, Deletions, Rearrangements & Changes of one or more Trunks to existing Trunk Groups:			
Per Interface, Occasion or Trunk Group	\$ 42.50	\$ 52.50	

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112

6.9 Message Telecommunications Service

6.9.1 Switched Outbound Service

6.9.1.1 IntraLATA Switched Outbound Service (see Section 10.8.1 for Price List Rates)

IntraLATA Switched Outbound Service is billed in six (6) second increments within initial billing period of eighteen (18) seconds. A total number of minutes will be rounded at end of billing period to the nearest minute. Fractional cents will be rounded to nearest cent.

Ameritech Serving Area:

<u>Usage</u>	Pea	<u>k</u>	<u>Off-</u>	Peak 199
	$\frac{\text{Min}}{10.04}$	<u>Max</u>	Min food	$\frac{Max}{0.12}$
Analog Service	\$0.04	\$0.13	\$0.04	\$0.13
Digital Service	\$0.04	\$0.13	\$0.04	\$0.13
Peak	Monday - Frida	y	8:00am - 4:59	pm
Off-Peak	Monday - Frida	iy	5:00pm - 7:59	am
	Saturday, Sund	ay	All Day	
	Holidays	-	All Day	

All times are Central Standard or Central Daylight Savings Time.

Cincinnati Bell Serving Area:

Cincinnati Den Sei ving ritea.		
	Min	Max
All Service, 24 Hours Per	\$0.04	\$0.60
Day, 7 Days Per Week		

6.9.1.2 InterLATA/Intrastate Switched Outbound Service

InterLATA/Intrastate Switched Outbound Service is billed in six (6) second increments, with a six (6) second call minimum. After the initial 6 second minimum, each fractional call is rounded up to the next one-tenth minute. Customers will receive the prices below 24 hours a day, 7 days a week.

<u>Usage</u>	<u>Minimum</u>	<u>Maximum</u>
Outbound Usage	\$0.04 per minute	\$0.13 per minute
On-Switch Customer Discount (applicable on per minute usage charges)	10%	

Issued: September 1, 1999

Effective: September 1, 1999

(R)

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

6.9 Message Telecommunications Service (cont'd)

6.9.2 Switched Inbound Service (ICG Toll Free) (see Section 10.8.2 for Price List Rates).

6.9.2.1 IntraLATA Switched Inbound Service

IntraLATA Switched Inbound Service is billed in six (6) second increments, with a six (6) second call minimum. After the initial 6 second minimum, each fractional call is rounded up to the next one-tenth minute. Customers will receive the prices below 24 hours a day, 7 days a week.

Usage	Min	<u>Max</u>
Inbound Usage	\$0.04 per minute	\$0.13 per minute
Per-Call Payphone Service Charge:	\$0.142	\$0.284
On-Switch Customer Discount (applicable on per minute usage charges)	10%	
(applicable on per innuce usage charges)	1070	

6.9.2.2 InterLATA/Intrastate Switched Inbound Service

InterLATA/Intrastate Switched Inbound Service is billed in six (6) second increments, with a six (6) second call minimum. After the initial 6 second minimum, each fractional call is rounded up to the next one-tenth minute. Customers will receive the prices below 24 hours a day, 7 days a week.

Usage	<u>Min</u>	Max
Inbound Usage	\$0.04 per minute	\$0.13 per minute
Per-Call Payphone Service Charge:	\$0.142	\$0.284
On-Switch Customer Discount (applicable on per minute usage charges)	10%	

6.9.3 MTS Services - Plan B

Plan B covers all instate long distance services (IntraLATA/Intrastate, InterLATA/Intrastate, outbound and inbound calls) with one rate which is applied to all usage. In order to qualify for Plan B service, customers must sign a minimum one year term agreement for ICG long distance service, and select ICG as their primary interexchange carrier for both InterLATA and IntraLATA long distance. All service is billed in six (6) second increments with a six (6) second call minimum.

	Min	Max	
Usage	\$0.025	\$0.100	
Per-Call Payphone Service Charge (inbound only)	\$0.142	\$0.284	

6.10 <u>Business Custom Calling Features, Ameritech Serving Area</u>: (see Section 10.9 of this tariff for Price List Rates).

Feature Description	Monthly Recu	rring Charges	Non Recurri	Non Recurring Charges		
	<u>Minimum</u>	<u>Maximum</u>	<u>Minimum</u>	<u>Maximum</u>		
Hunting	\$2.55	\$3.15	\$1.70	\$2.10		
Call Forwarding	\$2.97	\$3.68	\$6.29	\$7.77		
Call Forwarding Busy	\$2.97	\$3.68	\$6.29	\$7.77		
Call Forwarding No Answer	\$2.97	\$3.68	\$6.29	\$7.77		
Call Waiting	\$5.52	\$6.83	\$6.29	\$7.77		
3-Way Conference Calling	\$2.97	\$3.68	\$6.29	\$7.77		
Speed Call (8)	\$2.97	\$3.68	\$6.29	\$7.77		
Speed Call (30)	\$2.97	\$3.68	\$6.29	\$7.77		
Caller ID(w/o name)	\$6.37	\$7.88	\$6.29	\$7.77		
Caller ID(with name)	\$4.25	\$17.00	\$3.70	\$14.80		
Retained # (DNCF)						
with 4 Paths	\$3.40	\$4.20	\$42.50	\$52.50		
Each Additional Path	\$0.85	\$1.15	\$0.00	\$0.00		
Vanity Number	\$0.00	\$0.00	\$27.54	\$34.02		
Touch Tone	\$0.00	\$0.00	\$0.00	\$0.00		
911 Location Manager					(N)	
- Per DS0 for Digital Trunks	\$45.00	\$72.00	\$40.00	\$60.00	(N)	
- Per PRI	\$45.00	\$72.00	\$40.00	\$60.00	(N)	

Issued: July 12, 2002

Effective: August 12, 2002

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

6.0

NETWORK SERVICES - RETAIL PRICE LIST (cont'd)

Business Custom Calling Features, Cincinnati Bell Serving Area:**

Feature Description	Monthly Recur Minimum	ring Charges Maximum
Call Waiting	\$1.13	\$18.00
3-Way Calling	\$0.60	\$ 9.60
Call Forward (Variable)	\$0.60	\$ 9.60
Call Forward (Busy)	\$0.20	\$ 3.20
Call Forward (No Answer)	\$0.20	\$ 3.20
Speed Call - 8	\$0.68	\$10.80
Speed Call - 30	\$0.68	\$10.80
Caller ID (Without Name)	\$1.81	\$29.00
Caller ID (With Name)	\$4.25	\$17.00
Hunting	\$1.63	\$26.00
*DNCF - Listed #'s (up to 4 paths)	\$1.00	\$29.00
*DNCF - Unlisted #'s (up to 4 paths)	\$7.25	\$29.00
*DNCF - Each Additional Path	\$1.00	\$ 2.00
*Touch Tone	\$1.00	\$29.00
911 Location Manager		
- Per DS0 for Digital Trunks	\$45.00	\$72.00
- Per PRI	\$45.00	\$72.00

(N) (N)

N)

Non Recurring Charges	<u>Minimum</u>	Maximum	
Feature Installation, per order, per location	\$ 1.00	\$ 60.00	(I)

*Non Recurring charges do not apply

**The Company may waive these charges by publishing a Tariff Price List reflecting no charge.

Issued: July 12, 2002

Effective: August 12, 2002

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

ICG Telecom Group, Inc. 161 Inverness Drive West Englewood, CO 80112

 6.0
 NETWORK SERVICES - RETAIL PRICE LIST (cont'd)

 6.10
 Business Custom Calling Features, Ameritech Serving Area (cont'd)/see Section 10.9 tariff for Price List Rates).

The following Custom Calling Features are available for Volume Discount:*

- Call Forwarding
- Call Forwarding Busy
- Call Forwarding No Answer
- Call Waiting
- 3-Way Conference Calling
- Speed Call (8)
- Speed Call (30)

# Of Features Sold at 1 Time	Monthly Recurring Charge* Discounts
 Custom Calling Features per Line Custom Calling Features per Line Custom Calling Features per Line or more Custom Calling Features per 	20% 30% 35% Line 40%

* Custom calling feature discount is not available for service orders signed after November 5, 1999.

Business Custom Calling Features, Cincinnati Bell Serving Area**

The Cincinnati Bell Serving Area Custom Calling feature ranges listed on the preceding page of this tariff can be combined for a quantity discount. The quantity discount is on a per line, per month basis and the total discount amount for all lines will be deducted from the total monthly recurring charges per account for custom calling features.

	Monthly Recurring Discounts		
	Minimum	<u>Maximum</u>	
2 Custom Calling Features per Line	\$1.00	\$ 3.00	
3 Custom Calling Features per Line	\$1.00	\$ 4.00	
4 Custom Calling Features per Line	\$1.00	\$ 7.00	
5 or more Custom Calling Features per Line	\$1.00	\$11.80	

* Custom calling feature discount is not available for service orders signed after November 5, 1999.

**The Company may waive these charges by publishing a Tariff Price List reflecting no charge.

Issued: October 4, 1999

Effective: November 5, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 (T)

6.11 <u>900/976Blocking/Unblocking</u> (see Section 10.10 of this tariff for Price List Rates). Customers are given the option of per call blocking at no charge on a per line blocking at a rate listed below.

Ameritech Serving Area	<u>Minimum</u>	<u>Maximum</u>
Per Call Blocking	N/C	N/C
Per Line Blocking	N/C	N/C
<u>Cincinnati Bell Serving Area</u> **	<u>Minimum</u>	<u>Maximum</u>
Per Line Blocking	\$1.00	\$29.00

6.12 Miscellaneous

6.12.1 Maintenance Visit Charges (see Section 10.11 of this tariff for Price List Rates).

	Ŧ			
Ameritech Serving Area				
Duration of time, per technician		<u>First Hour</u>	Each Additional 15	
		.	<u>Minutes</u>	
Monday-Friday (8am-5pm)	Min	\$ 66.00	\$ 16.00	
	Max	\$184.00	\$ 46.00	
Monday-Friday (5pm-8am) & Saturday	Min	\$ 87.00	\$ 22.00	
	Max	\$184.00	\$ 46.00	
Sunday	Min	\$117.00	\$ 29.00	
	Max	\$184.00	\$ 46.00	
All ICG Recognized Holidays	Min	\$117.00	\$ 29.00	
	Max	\$184.00	\$ 46.00	
Trouble Isolation Charge (Flat Rate)	Min	\$ 35.00		
Û X X	Max	\$140.00		
Cincinnati Bell Serving Area				
Duration of time, per technician		<u>First Hour</u>	Each Additional 15	
- •			<u>Minutes</u>	
Monday-Friday (8am-5pm)	Min	\$ 66.00	\$ 16.00	
	Max	\$263.00	\$ 66.00	
Monday-Friday (5pm-8am) & Saturday	Min	\$ 87.00	\$ 22.00	
	Max	\$350.00	\$ 88.00	
Sunday	Min	\$117.00	\$ 29.00	
· · · · · · · · · · · · · · · · · · ·	Max	\$468.00	\$118.00	
All ICG Recognized Holidays	Min	\$117.00	\$ 29.00	
	Max	\$468.00	\$118.00	
Trouble Isolation Charge (Flat Rate)	Min	\$ 35.00		
	Max	\$140.00		

**The Company may waive these charges by publishing a Tariff Price List reflecting no charge.

Issued: February 19, 2001

Effective: March 1, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-249-TP-ATA

161 In	Celecom Group, Inc. Iverness Drive West wood, CO 80112		PUCO Tariff No. 2 Original Page 99.1	
6.0	NETWORK SERVICES - RETAIL PRICE LIS	$\underline{\Gamma}$ (cont'd)		
6.12	Miscellaneous (cont'd)			(N)
	6.12.2 Order Processing Charge			(N)
	Cincinnati Bell Service Area			(T)(M) (M)
	Non Recurring Charges	Minimum	<u>Maximum</u>	
	Per Location Per Service Order			
	- 24 or fewer lines/DS0's per change request, per service address	\$ 6.25	\$ 50.00	
	- 25 or more lines/DS0's per change request, per service address	\$10.00	\$ 80.00	
	Per PIC Code Change	\$ 1.25	\$ 10.00	 (M)
	Ameritech Service Area			(T)(M)
	Non Recurring Charges	<u>Minimum</u>	Maximum	(M)
	Per Location Per Service Order			ł
	- 24 or fewer lines/DS0's per change request, per service address	\$ 16.57	\$ 20.48	
	- 25 or more lines/DS0's per change request, per service address	\$10.00	\$ 80.00	
	Per PIC Code Change	\$ 1.25	\$ 10.00	 1
				1 (M)

*Material on this page was previously located on page 100.1.

Issued: February 19, 2001

Effective: March 1, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-249-TP-ATA Issued by: Amy Hartzler

Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112

6.14 <u>Calling Card Services</u> (see Section 10.13 of this tariff for Price List Rates). Minimum Maximum

	Minimum	iviaxinnum
Charge per minute	\$0.16/minute	\$0.30/minute
(within the continental US)		
Operator Assistance	••••••	\$0.68/call
Directory Assistance	\$0.00/call	\$0.74/call

The Company's Calling Card Services are offered to Customers of its local exchange (dial tone) services, such as Standard Business Line, and PBX Trunk Service. Calling Card calls are billed in six (6) second increments with a thirty (30) second call minimum. After the initial thirty (30) second minimum, each fractional call is rounded up to the nearest one-tenth of a minute.

The Company's Calling Card Services are accessed by dialing a toll-free number and entering the Calling Card Number (area code, phone number, and PIN).

6.15 Bad Check Charges

<u>Minimum</u>	<u>Maximum</u>
\$21.25	\$26.25

*Material on this page was previously located on page 100.

(C)

6.16 Digital Access Service

6.16.1 Rate Element Descriptions -- Standard Features (Plans A and B)

.1 DS1 Central Office Access Port. The port is used to connect the Digital Access Loop to the ICG serving switch. Monthly recurring and non-recurring installation charges apply per DS-1 ordered. Three access port options are available.

Option 1 (23B +Primary D Channel): ICG provisions 23, 64Kbps Bearer (B) channels and one 64Kbps Delta (D) channel over a 1.544Mbps DS1 transport facility.

- All 23 64Kbps Bearer channels must originate in one ICG switch and physically terminate within the same rate center as the Customer's physical location.
- Available only within calling areas served by ICG NPA-NXXs.
- May be combined with Options 2 and 3 to form a larger DAS Serving Arrangement, up to a maximum of 478 64Kbps B channels or 20 DS-1 facilities. Each DAS Serving Arrangement requires the purchase of at least one Option 1.
- Provides one rate center per DAS Serving Arrangement within a single ICG switch local serving area.

Option 2 (24B): ICG provisions 24, 64Kbps Bearer channels over a 1.544Mbps DS1 transport facility. The 24 Bearer (B) channels in Option 2 are controlled by the Delta (D) channel from Option 1.

- The customer may order DAS arrangements that combine an Option 1 configuration with multiple Option 2 configurations to form a larger trunk group (up to a maximum of 478 64Kbps B channels). ICG requires a minimum purchase of one Option 1 configuration per DAS arrangement; therefore, ICG will not provision Option 2 as a stand alone product.

Issued: September 1, 1998

Effective: October 1, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112 (N)

6.16 Digital Access Service (cont'd)

6.16.1 Rate Element Descriptions -- Standard Features (cont'd)

.1 DS1 Central Office Access Port (cont'd)

Option 2 (24B) (cont'd):

- All 24 64Kbps Bearer channels must originate in one ICG switch and physically terminate within the same rate center as the Customer's physical location.
- Provides one rate center per DAS Serving Arrangement within a single ICG switch local serving area.

Option 3 (23B + Backup D Channel): For redundancy purposes, the customer may purchase 23B and Backup Delta (D) channel which is provisioned over a separate T-1 facility from the 23B + Primary D Channel. Option 3 combines multiple digital access loops on a loop interface to function as one group.

- A minimum purchase of Option 1 is required per DAS Serving Arrangement; therefore, ICG will not provision Option 3 as a stand-alone product.
- .2 Digital Access Loop. The Digital Access Loop is a digital 1.544Mbps DS1 trunk facility ordered from the Customer Premises to the ICG network point of presence, or from the Customer collocation space within the ICG network point of presence. A Digital Access Loop must be purchased on every order associated with the purchase of a Central Office Port (Options 1, 2, and 3). Digital Access Loops are not for sale without Options 1, 2, or 3. Monthly recurring and non-recurring installation charges apply per DS1 ordered. The Digital Access Loop is offered only where facilities and operating conditions permit and is not offered across LATA boundaries. The Digital Access Loop charge will be waived if the Customer is collocated.

Issued: September 1, 1998

Effective: October 1, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112 (N)

6.16 Digital Access Service (cont'd)

6.16.1 Rate Element Descriptions -- Standard Features (cont'd)

- .3 **DAS Subgroup:** This feature allows customers who subscribe to multiple associated service types within a single DAS serving arrangement to create subgroups, thereby dedicating a certain number of channels to a particular service type. DAS subgroup is available with Plan B only.
- .4 **Caller Identification:** This feature displays the billing phone number of the calling party on specially designed customer-provided telephone equipment. Caller ID will only be provided when the phone number information is available to ICG. Caller ID will provide the phone number only.
- .5 Hunting within a DAS Subgroup/Arrangement: This feature provides the ability to send traffic from one DAS to another within the same DAS Service Arrangement. Hunting within a DAS Arrangement may be sequential or uniform call distribution. DAS Service Arrangements may only be comprised of DAS from one Plan (Plan A or Plan B).

Issued: September 1, 1998

Effective: October 1, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112

6.16 Digital Access Service (cont'd)

6.16.2 Rate Element Descriptions -- Optional Features

- .1 Called Number: this feature provides the ability for ICG's customer to prioritize inbound calls from their customers, based on the dialed digits. The customer may choose to designate multiple DAS subgroups within a single DAS Serving Arrangement, as facilities and operating conditions permit. The customer will be required to purchase DAS Subgroups and additional telephone numbers to implement the called number functionality.
- .2 Expanded Originating Service (EOS): EOS is available only with Plan A, and is an <u>inbound</u> Foreign Exchange-like digital dial tone service, using a DS1 transmission facility for delivery. EOS consists of two parts, and is subject to the following provisions:
 - ICG provides phone numbers (NPA-NXX's) for multiple calling areas within a switch service area. The customer may purchase phone numbers from calling areas within the same LATA as the physical customer's premises, or outside of the LATA that ICG has designed for this feature.
 - EOS also includes the ability to aggregate traffic from multiple calling areas within an ICG switch serving area. The aggregation of traffic from multiple calling areas may take place within the same LATA as the customer's premises or in another LATA that ICG has designed for this feature. Traffic from multiple ICG switch serving areas may not be aggregated within the same DAS Serving Arrangement. Separate DAS Serving Arrangements must be purchased per ICG switch.

(N)

Issued: September 1, 1998

Effective: October 1, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112

6.16 Digital Access Service (cont'd)

6.16.2 Rate Element Descriptions -- Optional Features (cont'd)

- .2 Expanded Originating Service (EOS) (cont'd)
 - Traffic from calling areas that are served by different ICG switches requires a minimum purchase of one Central Office Port and Digital Access Loop per switch serving area.
 - InterLATA EOS requires the purchase of EOS InterLATA transport components in addition to the EOS charges outlined in the Digital Access Tariff.
 - DID, if selected as an option, is necessary per local calling area requested by the Customer. Monthly recurring and non-recurring installation charges apply as specified below.
 - Because the hunting feature is provisioned as an inbound-only service, E-911 connectivity is unavailable; therefore, the customer is required to separately arrange for Local Exchange Service from ICG or another certified local exchange service provider to provide E-911 services.

Issued: September 1, 1998

Effective: October 1, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112 (N)

6.16 Digital Access Service (cont'd)

6.16.2 Rate Element Descriptions -- Optional Features (cont'd)

.2 Expanded Originating Service (EOS) (cont'd)

Routing Establishment Charge: a charge assessed per DS1 facility for configuring access from multiple calling areas that are served by a single ICG switch. Monthly recurring and non- recurring charges apply per DS1 that is ordered by the Customer.

EOS InterLATA Transport Charge: a charge assessed per DS1 facility for transporting traffic from one ICG switch to another between two separate LATAs. Monthly recurring and non-recurring charges apply per DS1.

(N)

(N)

Issued: September 1, 1998

Effective: October 1, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112

6.16 Digital Access Service (cont'd)

6.16.2 Rate Element Descriptions -- Optional Features (cont'd)

.3 Overflow Among DAS Arrangements: provides the ability to send traffic from one DAS Service Arrangement to other when the previous DAS Service Arrangement is full. In order to implement Overflow among DAS Service Arrangements, the customer must have more than one DAS Service Arrangement, each designated by a separate telephone number. Overflow among DAS Arrangements can only occur among DAS Service Arrangement with the same Plan (A or B) type. This feature will be available as facilities and operating conditions permit.

Monthly Recurring Charge and Non-Recurring Charges apply per overflow to each DAS Service Arrangement. For example, a customer ordering overflow among three DAS service arrangements will be assessed two overflow charges.

- .4 **DID:** enables incoming calls to be completed to CPE without attendant intervention through the out-pulsing of digits. Charges per number block apply in full, regardless of the number of DID numbers actually utilized by the Customer.
- .5 Number Block Charges: provides the ability for the Customer to order additional telephone numbers. Charged per block of 20 phone numbers. There will be no ability to reserve numbers for future use except by purchasing number blocks (subject to availability).
- .6 Additional Trunk Phone Number: ICG will provide one (1) phone number per DAS Serving Arrangement. Individual telephone numbers are available for purchase on a per number basis as facilities and operating conditions permit.
- .7 Change: Changes relate to the reconfiguration of existing service and/or the modification of any network component. A change is per customer request and per service address. There are two pricing components involved within a change request.

1-24 = a request of 24 lines (DS-0) or less. 25+ = a request of 25 lines (DS-0) or more.

There is a separate charge for each component respectively.

.8 Redirected Number Delivery: Enables the Customer to "trap" a number that has been call forwarded during the calling process using redirection information for further call | (N) | (N) | (N)

Issued: June 1, 1999

Effective: July 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: Julia Waysdorf Vice President, Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 ICG Telecom Group, Inc. 161 Inverness Drive West Englewood, CO 80112

6.0 <u>NETWORK SERVICES - RETAIL PRICE LIST</u> (cont'd)

6.16 Digital Access Service (cont'd)

Rates and Charges (Ameritech Serving Area): (Per DS-1, unless indicated otherwise)

Term / Rate Element	Ī	Monthly	1 Year	2 Year	3 Year	4 Year	5 Year	NRC*
PLAN A Central Office Port	MIN	\$ 209.25	\$ 202.50	\$ 186.75	\$ 168.75	\$ 146.75	\$ 140.00	\$ 475.00
Option 1	MAX	\$ 837.00	\$ 810.00	\$ 746.00	\$ 675.00	\$ 587.00	\$ 560.00	\$1900.00
PLAN A Central Office Port	MIN	\$ 188.00	\$ 182.25	\$ 168.00	\$ 151.50	\$ 139.00	\$ 133.00	\$ 475.00
Option 2	MAX	\$ 752.00	\$ 729.00	\$ 672.00	\$ 606.00	\$ 556.00	\$ 532.00	\$1900.00
PLAN A Central Office Port	MIN	\$ 265.00	\$ 256.50	\$ 236.50	\$ 213.75	\$154.00	\$ 147.00	\$ 475.00
Option 3	MAX	\$1060.00	\$ 1026.00	\$ 946.00	\$ 855.00	\$616.00	\$ 588.00	\$1900.00
PLAN A Digital Access	MIN	\$ 75.00	\$ 75.00	\$ 75.00	\$ 75.00	\$ 75.00	\$ 75.00	\$ 250.00
Loop**	MAX	\$ 300.00	\$ 300.00	\$ 300.00	\$ 300.00	\$ 300.00	\$ 300.00	\$1000.00
PLAN B Central Office Port	MIN	\$ 188.00	\$ 182.25	\$ 168.00	\$ 151.50	\$ 132.50	\$ 126.75	\$ 475.00
Option 1	MAX	\$ 752.00	\$ 729.00	\$ 672.00	\$ 606.00	\$ 530.00	\$ 507.00	\$1900.00
PLAN B Central Office Port	MIN	\$ 169.00	\$ 164.00	\$ 151.00	\$ 136.50	\$ 126.00	\$ 120.00	\$ 475.00
Option 2	MAX	\$ 676.00	\$ 656.00	\$ 604.00	\$ 546.00	\$ 504.00	\$ 480.00	\$1900.00
PLAN B Central Office Port	MIN	\$ 209.25	\$ 202.50	\$ 186.75	\$ 168.75	\$ 139.00	\$ 266.00	\$ 475.00
Option 3	MAX	\$ 837.00	\$ 810.00	\$ 747.00	\$ 675.00	\$ 556.00	\$ 532.00	\$1900.00
PLAN B Digital Access Loop	MIN	\$ 260.00	\$ 254.50	\$ 251.50	\$ 243.00	\$ 240.00	\$ 234.00	\$ 250.00
	MAX	\$1040.00	\$ 1018.00	\$1006.00	\$ 972.00	\$ 960.00	\$ 936.00	\$1000.00
- DAS Subgroup (PLAN B only) - Caller Identification - Hunting Within a DAS Subgroup/Arrangement	MIN MAX	\$ 0.00 \$ 50.00						
Called Number	MIN	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00
	MAX	\$ 150.00	\$ 150.00	\$ 150.0	\$ 150.00	\$ 150.00	\$ 150.00	\$ 150.00
Overflow Among DAS	MIN	\$ 10.00	\$ 10.00	\$ 10.00	\$ 10.00	\$ 10.00	\$ 10.00	\$ 75.00
Arrangements	MAX	\$ 40.00	\$ 40.00	\$ 40.00	\$ 40.00	\$ 40.00	\$ 40.00	\$ 300.00
Number Blocks (per block of 20 numbers)	MIN	\$ 1.38	\$ 1.38	\$ 1.38	\$ 1.38	\$ 1.38	\$ 1.38	\$ 50.00
	MAX	\$ 5.50	\$ 5.50	\$ 5.50	\$ 5.50	\$ 5.50	\$ 5.50	\$ 200.00
Redirected Number Delivery	MIN	\$12.50	\$12.50	\$12.50	\$12.50	\$12.50	\$12.50	\$50.00
	MAX	\$50.00	\$50.00	\$50.00	\$50.00	\$50.00	\$50.00	\$200.00

*NRC - Non-recurring charge

**Digital Access Loop charge for Plan A will be waived if customer is collocated. All rates may be pro-rated for a partial month.

Issued: June 1, 1999

Effective: July 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: Julia Waysdorf Vice President, Government Affairs 161 Inverness Drive West Englewood, Colorado 80112

6.16 Digital Access Service (cont'd) Rates and Charges (Ameritech Serving Area): (Per DS-1, unless indicated otherwise)

Term / Rate Element		Monthly	1 Year	2 Year	3 Year	4 Year	5 Year	NRC*
Additional Trunk Phone Number	MIN MAX	\$ 2.50 \$ 10.00	\$ 2.50 \$ 10.00	\$ 2.50 \$ 10,00	\$ 2.50 \$ 10.00	\$ 2.50 \$ 10.00	\$ 2.50 \$ 10.00	\$ 0.00 \$ 50.00
Changes - 1-24 Channels - 25+ Channels	MIN MAX MIN MAX	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00	\$ 12.50 \$ 50.00 \$ 20.00 \$ 80.00
EOS IntraLATA transport Cleveland Only	MIN MAX	\$ 62.00 \$ 250.00	\$ 62.00 \$ 250.00	\$ 62.00 \$ 250.00	\$ 62.00 \$ 250.00	N/A	N/A	\$200.00 \$800.00
Expanded Originating Service	MIN	\$ 12.50	\$ 12.50	\$ 12.50	\$ 12.50	\$ 12.50	\$ 12.50	\$ 75.00
 Routing Establishment Charge (per DAS) 	MAX	\$ 50.00	\$ 50.00	\$ 50.00	\$ 50.00	\$ 50.00	\$ 50.00	\$300.00
Expanded Originating Service - InterLATA Transport Charge	MIN	\$ 100.00	\$ 100.00	\$100.00	\$ 100.00	\$ 100.00	\$ 100.00	\$200.00
Route:Akron MSA- Cleveland MSA	МАХ	\$ 400.00	\$ 400.00	\$400.00	\$ 400.00	\$ 400.00	\$ 400.00	\$800.00
Akron MSA - Columbus MSA	MIN MAX	\$ 220.00 \$ 880.00	\$ 220.00 \$ 880.00	\$220.00 \$880.00	\$ 220.00 \$ 880.00	\$ 220.00 \$ 880.00	\$ 220.00 \$ 880.00	\$200.00 \$800.00
Akron MSA - Dayton MSA	MIN MAX	\$ 340.00 \$ 1360.00	\$ 340.00 \$1360.00	\$ 200.00 \$ 800.00				
Cleveland MSA - Columbus MSA	MIN MAX	\$ 255.00 \$ 1020.00	\$ 200.00 \$ 800.00					
Cleveland MSA - Dayton MSA	MIN MAX	\$ 360.00 \$ 1440.00	\$ 200.00 \$ 800.00					
Columbus MSA - Dayton MSA	MIN MAX	\$ 130.00 \$ 520.00	\$ 200.00 \$ 800.00					

Term/Rate Element		2 Year	3 Year	4 Year	5 Year
Plan C - Central Office Port	MIN	16%	14%	15%	17%
Option 1	MAX	40%	40%	40%	45%
Plan C - Digital Access Loop	MIN	2%	3%	5%	6%
Option 1	MAX	10%	14%	20%	24%

(T) (T)

Discounts based on Plan A, Option 1 pricing. 4 and 5 year term discounts based on Plan A, Option 1 pricing.

4 and 5 year term discounts based on Plan A, Option 1 pr Please refer to Plan A for all applicable NRC's.

*NRC - Non-recurring charge

All rates may be pro-rated for a partial month.

Issued: July 2, 1999

Effective: August 2, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-1650-TP-TRF.

Issued by: Julia Waysdorf Vice President, Government Affairs 161 Inverness Drive West Englewood, Colorado 80112

6.16 Digital Access Service (cont'd) Botos and Charges (Ameritash Serving A

Term/Rate Elem	ent	2 Year	3 Year	4 Year	5 Year	NRC*
Plan C - DS3 EOS InterL. Transport Charge Route:	ATA					
Akron to Cleveland	Min	\$950.00	\$950.00	\$950.00	\$950.00	\$500.00
	Max	\$3,800.00	\$3,800.00	\$3,800.00	\$3,800.00	\$2,000.00
Akron to Columbus	Min	\$1,437.00	\$1,437.00	\$1,437.00	\$1,437.00	\$500.00
	Max	\$5,750.00	\$5,750.00	\$5,750.00	\$5,750.00	\$2,000.00
Akron to Dayton	Min	\$2,200.00	\$2,200.00	\$2,200.00	\$2,200.00	\$500.00
	Max	\$8,800.00	\$8,800.00	\$8,800.00	\$8,800.00	\$2,000.00
Cleveland to Columbus	Min	\$1,650.00	\$1,650.00	\$1,650.00	\$1,650.00	\$500.00
	Max	\$6,600.00	\$6,600.00	\$6,600.00	\$6,600.00	\$2,000.00
Cleveland to Dayton	Min	\$2,325.00	\$2,325.00	\$2,325.00	\$2,325.00	\$500.00
	Max	\$9,300.00	\$9,300.00	\$9,300.00	\$9,300.00	\$2,000.00
Columbus to Dayton	Min	\$950.00	\$950.00	\$950.00	\$950.00	\$500.00
	Max	\$3,800.00	\$3,800.00	\$3,800.00	\$3,800.00	\$2,000.00

Rates and Charges (Ameritech Serving Area):

*NRC - Non-Recurring Charge

All rates may be pro-rated for a partial month.

Term / Rate Element		3 Year	4 Year	5 Year
Plan C-Central Office Port Option 2 (250-499 PRI)	Min Max	\$200.00 \$800.00	\$200.00 \$800.00	\$200.00 \$800.00
(500 + PRI)	Min Max	\$175.00 \$700.00	\$175.00 \$700.00	\$175.00 \$700.00

-Option 2 includes pricing for Central Office Port Only.

-All other DAS features and options are at the rates found previously in this section. -Please refer to Plan A for all applicable NRC's.

Issued: July 2, 1999

Effective: August 2, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-1650-TP-TRF.

Issued by: Julia Waysdorf Vice President, Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 -----

(N)

6.16 Digital Access Service (cont'd)

Rates and Charges (Ameritech Serving Area):

Term/Rate Elem	ent	Monthly	1 Year	2 Year	3 Year	NRC*
Plan D - DS3	Min	\$750.00	\$750.00	\$750.00	\$750.00	\$250.00
Digital Access Loop	Max	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00	\$1,000.00
Plan D - DS3 EOS IntraLATA transport Cleveland Only	Min Max	\$625.00 \$2,500.00	\$625.00 \$2,500.00	\$625.00 \$2,500.00	\$625.00 \$2,500.00	\$200.00 \$800.00
Plan D - EOS InterLA' Transport Route:	ΓA					
Akron to Cleveland	Min	\$1150.00	\$1150.00	\$1150.00	\$1150.00	\$250.00
	Max	\$4,600.00	\$4,600.00	\$4,600.00	\$4,600.00	\$1,000.00
Akron to Columbus	Min	\$1,980.00	\$1,980.00	\$1,980.00	\$1,980.00	\$250.00
	Max	\$7,920.00	\$7,920.00	\$7,920.00	\$7,920.00	\$1,000.00
Akron to Dayton	Min	\$3,060.00	\$3,060.00	\$3,060.00	\$3,060.00	\$250.00
	Max	\$12,240.00	\$12,240.00	\$12,240.00	\$12,240.00	\$1,000.00
Cleveland to	Min	\$2,295.00	\$2,295.00	\$2,295.00	\$2,295.00	\$250.00
Columbus	Max	\$9,180.00	\$9,180.00	\$9,180.00	\$9,180.00	\$1,000.00
Cleveland to Dayton	Min	\$3,240.00	\$3,240.00	\$3,240.00	\$3,240.00	\$250.00
	Max	\$12,960.00	\$12,960.00	\$12,960.00	\$12,960.00	\$1,000.00
Columbus to Dayton	Min	\$1,170.00	\$1,170.00	\$1,170.00	\$1,170.00	\$250.00
	Max	\$4,680.00	\$4,680.00	\$4,680.00	\$4,680.00	\$1,000.00

- DS-3 pricing replaces the Digital Access Loop, EOS IntraLATA transport, and/or EOS InterLATA transport commonly provisioned at the DS-1 level and available in requested markets.

- All other Digital Access Services pricing still apply. Please refer to pricing in Plan A, Plan B or Plan C for all other applicable charges.

*NRC - Non-Recurring Charge All rates may be pro-rated for a partial month.

Effective: April 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-1650-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112 (N)

6.16 Digital Access Service (cont'd)

Rates and Charges (Ameritech Serving Area):

Plan E - Cent	ral Office Port	Monthly	1 Year	2 Year	3 Year	4 Year	5 Year
25-49 PRI's	Min	\$230.00	\$220.00	\$196.00	\$184.00	\$173.00	\$162.00
	Max	\$920.00	\$880.00	\$784.00	\$736.00	\$692.00	\$650.00
50-99 PRI's	Min	\$230.00	\$220.00	\$192.00	\$180.00	\$169.00	\$162.00
	Max	\$920.00	\$880.00	\$768.00	\$720.00	\$676.00	\$650.00
100+ PRI's	Min	\$230.00	\$220.00	\$188.00	\$176.00	\$166.00	\$159.00
	Max	\$920.00	\$880.00	\$752.00	\$706.00	\$644.00	\$638.00

-Plane E includes Central Office Port only.

-All other DAS features and options are available at the rates found previously in this section. -Please refer to Plan A for all applicable NRC's.

Plan F - Central Office Port: 5 Year Term Commitment		<u>NRC</u> *	MRC	(T) (M)
200,000 + PRIs	Min Max		\$137.00** \$600.00**	

* -Please refer to Plan A For all applicable NRC's.

** -pricing available only for agreements signed prior to January 4, 2002 -Plan F includes Central Office Port only.

-All other DAS features and options are available at the rates found previously in this section.

Plan G-Central Office Port:		NRC	MRC
66 Month Term Commitment			
100,000 + PRI's			
Months 1-36	Min Max	\$0.00 \$500.00	\$256.00 \$1016.00
Months 37-66	Min Max	\$0.00 \$500.00	\$144.00 \$576.00

-Plan G includes Central Office Port Only

-All other DAS features and options are available at rates found previously in this section

Issued: December 4, 2001

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 Effective: January 4, 2002

(M)

⁽N)

Digital Access Service (cont'd) .16

<u>Rates and Charges</u>: (Per DS1, unless indicated otherwise) (Applies to both Ameritech and Cincinnati Bell Serving Areas)

PLAN B	n Ameritech and High Volume	1 Year	2 Year	3 Year	5 Year	Overage Per MOU*
<u>Term/Ra</u>	te Element					<u> </u>
<u>50,000 MO</u>	<u>U*</u>					
Overage	Minimum					\$0.0037
Overage	Maximum					\$0.0150
MRC**	Minimum	\$400.00	\$400.00	\$400.00	\$400.00	
	Maximum	\$1,600.00	\$1,600.00	\$1,600.00	\$1,600.00	
NRC***	<u>Minimum</u>	\$0.00	\$0.00	\$0.00	\$0.00	
	Maximum	\$1,100.00	\$1,100.00	\$1,100.00	\$1,100.00	
100,000 M	0 <u>U*</u>					
Overage	Minimum					\$0.0032
Overage	Maximum					\$0.0130
MRC**	Minimum	\$425.00	\$425.00	\$425.00	\$425.00	
	Maximum	\$1,700.00	\$1,700.00	\$1,700.00	\$1,700.00	
NRC***	Minimum	\$0.00	\$0.00	\$0.00	\$0.00	
	Maximum	\$1,100.00	\$1,100.00	\$1,100.00	\$1,100.00	
250,000 M	<u>OU*</u>					
Overage	Minimum					\$0.0027
Overage	Maximum	_				\$0.0110
MRC**	Minimum	\$500.00	\$500.00	\$500.00	\$500.00	1
	Maximum	\$2,000.00	\$2,000.00	\$2,000.00	\$2,000.00	
NRC***	Minimum	\$0.00	\$0.00	\$0.00	\$0.00	
	Maximum	\$1,100.00	\$1,100.00	\$1,100.00	\$1,100.00	
350,000 M	0 <u>U*</u>					
Overage	Minimum					\$0.0022
Overage	Maximum					\$0.009
MRC**	Minimum	\$575.00	\$575.00	\$575.00	\$575.00	
	Maximum	\$2,300.00	\$2,300.00	\$2,300.00	\$2,300.00	
NRC***	Minimum	\$0.00	\$0.00	\$0.00	\$0.00	
	Maximum	\$1,100.00	\$1,100.00	\$1,100.00	\$1,100.00	

- Charges for moves, charges and deletes found in this tariff also apply.

- Plan B High Volume rate includes central office port only; Digital Access Loop Charge waived if customer is collocated.

- Customer must still select Plan B configuration option 1, 2, or 3.

MOU - Minutes of Use MRC - Monthly Recurring Charge

NRC - Non-recurring charge

All rates may be pro-rated for a partial month.

Issued: August 24, 2001 Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. ____-TP-ATA

Effective: September 24, 2001

6.16 Digital Access Service (cont'd)

Rates and Charges (Cincinnati Bell Serving Area): (Per DS-1, unless indicated otherwise)

Term / Rate Element		Monthly	1 Year	2 Year	3 Year	4 Year	5 Year	NRC*
PLAN A Central Office Port	MIN	\$ 375.00	\$ 362.50	\$ 325.00	\$ 311.00	\$ 307.50	\$ 297.50	\$ 200.800
Option 1	MAX	\$ 1500.00	\$ 1450.00	\$ 1300.00	\$ 1244.00	\$ 1230.00	\$ 1190.00	\$ 800.00
PLAN A Central Office Port	MIN	\$ 356.00	\$ 344.50	\$ 309.00	\$ 295.50	\$ 292.50	\$ 283.00	\$ 200.800
Option 2	MAX	\$ 1426.00	\$ 1378.00	\$ 1236.00	\$ 1182.00	\$ 1170.00	\$ 1132.00	\$ 800.00
PLAN A Central Office Port	MIN	\$ 393.50	\$ 380.50	\$ 341.00	\$ 326.50	\$ 322.50	\$ 312.00	\$ 200.800
Option 3	MAX	\$ 1574.00	\$ 1522.00	\$ 1364.00	\$ 1306.00	\$ 1290.00	\$ 1248.00	\$ 800.00
PLAN A Digital Access	MIN	\$ 75.00	\$ 75.00	\$ 75.00	\$ 75.00	\$ 75.00	\$ 75.00	\$ 100.00
Loop**	MAX	\$ 300.00	\$ 300.00	\$ 300.00	\$ 300.00	\$ 300.00	\$ 300.00	\$ 400.00
PLAN B Central Office Port	MIN	\$ 387.50	\$ 375.00	\$ 350.00	\$ 340.00	\$ 326.00	\$ 307.50	\$ 200.800
Option 1	MAX	\$ 1550.00	\$ 1500.00	\$ 1400.00	\$ 1360.00	\$ 1304.00	\$ 1230.00	\$ 800.00
PLAN B Central Office Port	MIN	\$ 368.50	\$ 356.50	\$ 367.50	\$ 323.00	\$ 310.00	\$ 292.50	\$ 200.800
Option 2	MAX	\$ 1474.00	\$ 1426.00	\$ 1470.00	\$ 1292.00	\$ 1240.00	\$ 1170.00	\$ 800.00
PLAN B Central Office Port	MIN	\$ 406.50	\$ 393.50	\$ 332.50	\$ 357.00	\$ 342.00	\$ 322.50	\$ 200.800
Option 3	MAX	\$ 1626.00	\$ 1574.00	\$ 1330.00	\$ 1428.00	\$ 1368.00	\$ 1290.00	\$ 800.00
PLAN B Digital Access Loop	MIN	\$ 75.00	\$ 75.00	\$ 75.00	\$ 75.00	\$ 75.00	\$ 75.00	\$ 100.00
	MAX	\$ 300.00	\$ 300.00	\$ 300.00	\$ 300.00	\$ 300.00	\$ 300.00	\$ 400.00
- DAS Subgroup (PLAN B only) - Caller Identification - Hunting Within a DAS Subgroup/Arrangement	MIN MAX	no charge \$ 50.00						

*NRC - Non-recurring charge

**Digital Access Loop charge for Plan A will be waived if customer is collocated. All rates may be pro-rated for a partial month.

(N)

Issued: October 21, 1998

Effective: October 21, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112

6.16 **Digital Access Service** (cont'd) Rates and Charges (Cincinnati Bell Serving Area): (Per DS-1, unless indicated otherwise)

Term / Rate Element		Monthly	1 Year	2 Year	3 Year	4 Year	5 Year	NRC*
Called Number	MIN	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00
	MAX	\$ 50.00	\$ 50.00	\$ 50.00	\$ 50.00	\$ 50.00	\$ 50.00	\$ 50.00
Overflow Among DAS	MIN	\$ 10.00	\$ 10.00	\$ 10.00	\$ 10.00	\$ 10.00	\$ 10.00	\$ 10.00
Arrangements	MAX	\$ 40.00	\$ 40.00	\$ 40.00	\$ 40.00	\$ 40.00	\$ 40.00	\$ 40.00
Number Blocks	MIN	\$ 1.88	\$ 1.88	\$ 1.88	\$ 1.88	\$ 1.88	\$ 1.88	\$ 93.33
1 st block of 20 numbers	MAX	\$ 750.00	\$ 750.00	\$ 750.00	\$ 750.00	\$ 750.00	\$ 750.00	\$ 373.32
Number Blocks	MIN	\$ 1.50	\$ 1.50	\$ 1.50	\$ 1.50	\$ 1.50	\$ 1.50	\$ 1.00
Additional block of 20 numbers	MAX	\$ 6.00	\$ 6.00	\$ 6.00	\$ 6.00	\$ 6.00	\$ 6.00	\$40.00
Number Blocks	MIN	\$ 1.00	\$ 1.00	\$ 1.00	\$ 1.00	\$ 1.00	\$ 1.00	\$ 1.00
1" block of 20 numbers over 100	MAX	\$60.00	\$60.00	\$60.00	\$60.00	\$60.00	\$60.00	\$40.00
Additional Trunk Phone Number	MIN	\$ 2.50	\$ 2.50	\$ 2.50	\$ 2.50	\$ 2.50	\$ 2.50	\$ 0.00
	MAX	\$ 10.00	\$ 10.00	\$ 10.00	\$ 10.00	\$ 10.00	\$ 10.00	\$ 50.00
Charges - 1-24 Channels - 25+ Channels	MIN MAX MIN MAX	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00	\$ 0.00	\$12.50 \$ 50.00 \$ 20.00 \$ 80.00
Expanded Originating Service - Routing Establishment Charge (per DAS)	MIN MAX	\$ 12.50 \$ 50.00	\$ 75.00 \$ 300.00					
Expanded Originating Service - InterLATA Transport Charge Route:	MIN MAX	TBD						
Redirected Number Delivery	MIN	\$ 12.50	\$ 12.50	\$ 12.50	\$ 12.50	\$ 12,50	\$ 12.50	\$ 50.00
	MAX	\$ 50.00	\$ 50.00	\$ 50.00	\$ 50.00	\$ 50.00	\$ 50.00	\$ 200.00

Term/Rate Element		2 Year	3 Year	4 Year	5 Year
Plan C - Central Office Port	MIN	0%	0%	0%	1%
Option 1	MAX	15%	15%	15%	15%
Plan C - Digital Access Loop	MIN	2%	3%	5%	6%
Option 1	MAX	10%	14%	20%	24%

Discounts based on Plan A, Option 1 pricing. 4 and 5 year term discounts based on Plan A, Option 1 pricing. Please refer to Plan A for all applicable NRC's.

*NRC - Non-recurring charge

**Digital Access Loop charge for Plan A will be waived if customer is collocated.

All rates may be pro-rated for a partial month.

Effective: October 13, 1999 Issued: September 13, 1999 Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-1650-TP-TRF.

> Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112

(T)

(T)

6.16 Digital Access Service (cont'd) Rates and Charges (Cincinnati Bell Serving Area):

Term / Rate Element	[3 Year	4 Year	5 Year
Plan C-Central Office Port Option 2 (250-499 PRI)	Min Max	\$325.00 \$1300.00	\$325.00 \$1300.00	\$325.00 \$1300.00
(500+ PRI)	Min Max	\$300.00 \$1200.00	\$300.00 \$1200.00	\$300.00 \$1200.00

-Option 2 includes pricing for Central Office Port Only.

-All other DAS features and options are at the rates found previously in this section.

-Please refer to Plan A for all applicable NRC's.

Term/Rate Element	····	Monthly	1 Year	2 Year	3 Year	NRC*
Plan D - DS-3	MIN	\$750.00	\$750.00	\$750.00	\$750.00	\$250.00
Digital Access Loop	МАХ	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00	\$1,000.00

- DS-3 pricing replaces the Digital Access Loop, EOS IntraLATA transport, and/or EOS InterLATA transport commonly provisioned at the DS-1 level and available in requested markets.

- All other Digital Access Services pricing still apply. Please refer to pricing in Plan A, Plan B or Plan C for all other applicable charges. *NRC - Non-Recurring Charge

All rates may be pro-rated for a partial month.

Plan E - Cen	tral Office Port	Monthly	1 Year	2 Year	3 Year	4 Year	5 Year
25-49 PRI's	Min	\$375.00	\$362.00	\$318.00	\$308.00	\$305.00	\$298.00
	Max	\$1500.00	\$1450.00	\$1274.00	\$1232.00	\$1222.00	\$1194.00
50-99 PRI's	Min	\$375.00	\$362.00	\$312.00	\$305.00	\$302.00	\$296.00
	Max	\$1500.00	\$1450.00	\$1250.00	\$1220.00	\$1210.00	\$1184.00
100+ PRI's	Min	\$375.00	\$362.00	\$306.00	\$302.00	\$300.00	\$292.00
	Max	\$1500.00	\$1450.00	\$1224.00	\$1208.00	\$1200.00	\$1170.00

-Plan E includes Central Office Port only.

-All other DAS features and options are available at the rates found previously in this section.

-Please refer to Plan A for all applicable NRC's.

Issued: December 4, 2001

Effective: January 4, 2002

(N)

(N)

Original Page 101.11.2

6.0 NETWORK SERVICES - RETAIL PRICE LIST (cont'd)

Digital Access Service (cont'd) 6.16

Plan F - Central Office Port: 5 Year Term Commitment	<u>NRC</u> *	MRC
200,000 + PRIs		
Min		\$137.00**
Max		\$600.00**

* -Please refer to Plan A For all applicable NRC's. ** -pricing available only for agreements signed prior to January 4, 2002 -Plan F includes Central Office Port only.

-All other DAS features and options are available at the rates found previously in this section.

Plan G-Central Office Port: 66 Month Term Commitment	NRC	MRC
100,000 + PRI's		
Months 1-36		
Min	\$0.00	\$256.00
Max	\$500.00	\$1,016.00
Months 37-66		
Min	\$0.00	\$144.00
Max	\$500.00	\$576.00

-Plan G includes Central Office Port Only

-All other DAS features and options are available at rates found previously in this section

Issued: December 4, 2001

Effective: January 4, 2002

6.17 Business Line Connect

Ameritech Saving Area - Non-Recurring Charges Per Line

First Line	Min Max	\$ 37.00 \$150.00
Each Additional Line	Min Max	\$ 25.00 \$100.00

Monthly Recurring Charges

Term Plan		Month to Month	1 Year	2 Year	3 Year
Business Line	Min	\$19.00	\$18.00	\$17.00	\$16.00
Connect*	Max	\$76.00	\$72.00	\$69.00	\$65.00
Business Line	Min	\$22.00	\$21.00	\$20.00	\$19.00
Connect Plus**	Max	\$90.00	\$86.00	\$82.00	\$72.00

Cincinnati Bell Saving Area - Non-Recurring Charges Per Line

First Line	Min Max	\$ 37.00 \$150.00
Each Additional Line	Min Max	\$ 25.00 \$100.00

Monthly Recurring Charges

Term Plan		Month to Month	1 Year	2 Year	3 Year
Business Line	Min	\$17.00	\$17.00	\$17.00	\$17.00
Connect*	Max	\$68.00	\$68.00	\$68.00	\$68.00
Business Line	Min	\$23.00	\$21.00	\$18.00	\$18.00
Connect Plus**	Max	\$96.00	\$84.00	\$75.00	\$72.00

*Monthly Recurring Charge for Business Line Connect includes the following features: Standard Business Line, Intercom Dialing, Hunting, 3-way Conference Calling, Consultation Hold, Touch Tone, Call Transfer External, Call Forwarding Variable, Call Forwarding Busy and Call Forwarding Don't Answer.

**Monthly Recurring Charge for Business Line Connect Plus includes all features of Business Line Connect and Voice Mail.

Effective: April 1, 1999

(N)

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-1650-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112

6.18 Dedicated Long Distance Service, Ameritech and Cincinnati Bell Serving Area

6.18.1 Plan A: IntraLATA/Intrastate Dedicated Outbound Service

IntraLATA/Intrastate Dedicated Outbound Service is billed in six (6) second increments, with an (R) initial minimum billing period of six (6) seconds. After the initial 6 second minimum, each fractional call is rounded up to the next one-tenth minute. Customer's will receive the prices below 24 hours a day, 7 days a week.

	Minimum	Maximum
Outbound Usage: Dedicated Customer Discount	\$0.046 10%	\$0.184 60%
Deuleuteu Custonier Distount	2010	

6.18.2 Plan A: InterLATA/Intrastate Dedicated Outbound Service

InterLATA/Intrastate Dedicated Outbound Service is billed in six (6) second increments, with an (R) initial minimum billing period of six (6) seconds. After the initial 6 second minimum, each fractional call is rounded up to the next one-tenth minute. Customer's will receive the prices below 24 hours a day, 7 days a week.

	Minimum	Maximum
Outbound Usage:	\$0.046	\$0.184
Dedicated Customer Discount	10%	60%

6.18.3 Plan A: IntraLATA/Intrastate Dedicated Inbound Service (Dedicated Toll Free)

IntraLATA/Intrastate Dedicated Inbound Service is billed in six (6) second increments, with an (R) initial minimum billing period of six (6) seconds. After the initial 6 second minimum, each fractional call is rounded up to the next one-tenth minute. Customer's will receive the prices below 24 hours a day, 7 days a week.

	<u>Minimum</u>	<u>Maximum</u>
Outbound Usage:	\$0.046	\$0.184
Dedicated Customer Discount	10%	60%
Per Call Payphone Surcharge*	\$0.142	\$0.284

6.18.4 Plan A: InterLATA/Intrastate Dedicated Inbound Service (Dedicated Toll Free)

InterLATA/Intrastate Dedicated Inbound Service is billed in six (6) second increments, with an (R) initial minimum billing period of six (6) seconds. After the initial 6 second minimum, each fractional call is rounded up to the next one-tenth minute. Customer's will receive the prices below 24 hours a day, 7 days a week.

	<u>Minimum</u>	<u>Maximum</u>
Outbound Usage:	\$0.046	\$0.184
Dedicated Customer Discount	10%	60%
Per Call Payphone Surcharge*	\$0.142	\$0.284
		NAT 1 No. of Contract

6.18.5	Plan A: Dedicated Long Distance Directory Assistance	<u>Minimum</u>	<u>Maximum</u>
	(NPA) 555-1212	\$0.20/call	\$0.80/call
· · · · · · · · · · · · · · · · · · ·	- Dissount door not combr		

*Dedicated Customer Discount does not apply.

Issued: September 1, 1999

Effective: September 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-1650-TP-TRF.

6.19 Business Essentials

<u>Component A</u> - includes one standard Business Line Plus line and the following features; hunting, 3 Way calling, call transfer, consultation hold, Caller ID, Call Waiting, Call Forwarding/Variable /Busy/No Answer.

		Ameritech Serving	Cincinnati Serving Area
Term		Monthly Recurring	Monthly Recurring
12	Min	\$18.50	\$21.00
Month	Max	\$74.00	\$84.00
24	Min	\$17.50	\$20.00
Month	Max	\$70.00	\$80.00
36	Min	\$16.50	\$19.00
Month	Max	\$66.00	\$76.00

-Customers may elect to decline any of the Custom Calling features included in the Business Essentials Bundle, however this will not impact the monthly recurring rate.

-Additional Custom Calling features are available at the rates listed in this tariff.

<u>Component B</u> - Per minute rate applies to all inbound and outbound IntraLATA/intrastate, InterLATA/intrastate and interstate calls. Volume Discounts apply on a per month, per account basis.

Ameritech Serving Area				
Term		1-2499 MOU	2500-4999 MOU	5000+ MOU
12 Month	Min	\$0.037	\$0.035	\$0.033
	Max	\$0.150	\$0.140	\$0.130
24 Month	Min	\$0.037	\$0.035	\$0.033
	Max	\$0.150	\$0.140	\$0.130
36 Month	Min	\$0.037	\$0.035	\$0.033
	Max	\$0.150	\$0.140	\$0.130

-Calling Cards and International calls are at the regular tariffed rates.

-This Service is billed in six (6) seconds increments, with an initial minimum billing period of six (6) seconds.

Cincinnati Bell Serving Area						
Term	T T	1-2499 MOU	2500-4999 MOU	5000+ MOU		
12 Month	Min	\$0.042	\$0.039	\$0.037		
	Max	\$0.170	\$0.160	\$0.150		
24 Month	Min	\$0.042	\$0.039	\$0.037		
	Max	\$0.170	\$0.160	\$0.150		
36 Month	Min	\$0.042	\$0.039	\$0.037		
	Max	\$0.170	\$0.160	\$0.150		

-Calling Cards and International calls are at the regular tariffed rates.

-This Service is billed in six (6) seconds increments, with an initial minimum billing period of six (6) seconds.

Other Charges: (Ameritech and Cincinnati Bell Serving Area)	Min	Max
Nonrecurring Installation Charger First Line	\$25.00	\$100.00
Each Additional Line	\$12.50	\$60.00
Each Additional Line	\$12.50	ψ00.00

Early Termination Penalty-----\$100.00

Issued: January 28, 2000 Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 (N)

(N)

PUCO Tariff No. 2

Original Page 101.14

6.20 Optional Calling Plans to ICG LOCAL service

6.20.1 ICG LOCAL service customers can subscribe to a toll calling plan and one of the calling card options defined in ICG's "Toll" offerings.

| | (N)

(N)

Issued by: Scott Beer Director, Industry &Corporate Affairs 161 Inverness Drive West Englewood, Colorado 80112

8.0 MISCELLANEOUS SERVICES

8.1 Busy Line Verify and Line Interrupt Service

8.1.1 Description

Upon request of a calling party the Company will verify a busy condition on a called line.

A) The operator will determine if the line is clear or in use and report to the calling party.

B) The operator will interrupt the call on the called line only if the calling party indicates an emergency and requests interruption.

8.1.2 <u>Regulations</u>

A) A charge will apply when:

- 1) The operator verifies that the line is busy with a call in progress.
- 2) The operator verifies that the line is available for incoming calls.

3) The operator verifies that the called number is busy with a call in progress and the Customer requests interruption. The operator will then interrupt the call, advising the called party the name of the calling party. A separate charge will apply for both verification and interruption.

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112

8.0 MISCELLANEOUS SERVICES (cont'd)

8.1 Busy Line Verify and Interrupt Services (cont'd)

8.1.2 Regulations (cont'd)

- B) No charge will apply when:
 - 1) When the calling party advises that the call is to or from an official public emergency agency.
- C) Busy Verification and Interrupt Service is furnished where and to the extent that facilities permit.
- D) The Customer shall identify and save the Company harmless against all claims that may arise from either party to the interrupted call or any person.

8.1.3 Rates

Call Processing

<u>Minimum</u> 0.40 Maximum 0.60

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112

8.0 MISCELLANEOUS SERVICES (cont'd)

8.2 <u>Restoration of Service</u>

8.2.1 Description

A restoration charge applies to the restoration of suspended service and facilities because of nonpayment of bills and is payable at the time that the restoration of the suspended service and facilities is arranged.

8.2.2 Rates - Ameritech Serving Area

Per occasion per location

\$20.00

Non-Recurring

Non-Recurring \$25.00

Rates - Cincinnati Bell Serving Area

Per occasion per location

Issued: February 19, 2001

Effective: March 1, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-249-TP-ATA

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112

(T)

(T)

(T)

9.0 SPECIAL ARRANGEMENTS

9.1 Special Construction

9.1.1 Basis for Charges

Where the Company furnishes a facility or service for which a rate or charge is not specified in the Company's tariffs, charges will be based on the costs incurred by the Company and may include: (1) non-recurring type charges; (2) recurring type charges; (3) termination liabilities; or (4) combinations thereof.

9.1.2 Basis for Cost Computation

The costs referred to in 9.1.1 preceding may include one or more of the following items to the extent they are applicable:

- A) cost installed of the facilities to be provided including estimated costs for the rearrangements of existing facilities. Cost installed includes:
 - 1) equipment and materials provided or used,
 - 2) engineering, labor and supervision,
 - 3) transportation, and
 - 4) rights of way;
- B) cost of maintenance;
- C) depreciation on the estimated cost installed of any facilities provided, based on the anticipated useful service life of the facilities with an appropriate allowance for the estimated net salvage;
- D) administration, taxes and uncollectible revenue on the basis of reasonable average costs for these items;

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112

9.0 <u>SPECIAL ARRANGEMENTS</u> (cont'd)

9.1 Special Construction (cont'd)

9.1.2 Basis for Cost Computation (cont'd)

- E) license preparation, processing and related fees;
- F) tariff preparation, processing and related fees;
- G) any other identifiable costs related to the facilities provided; or
- H) an amount for return and contingencies.

9.1.3 <u>Termination Liability</u>

To the extent that there is no other requirement for use by the Company, a termination liability may apply for facilities specially constructed at the request of the customer.

- A) The termination liability period is the estimated service life of the facilities provided.
- B) The amount of the maximum termination liability is equal to the estimated amounts for:
 - 1) Cost installed of the facilities provided including estimated costs for rearrangements of existing facilities and/or construction of new facilities as appropriate, less net salvage. Cost installed includes the cost of:
 - a) equipment and materials provided or used,
 - b) engineering, labor and supervision,
 - c) transportation, and
 - d) rights of way;

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

9.0 SPECIAL ARRANGEMENTS (cont'd)

9.1 Special Construction (cont'd)

9.1.3 Termination Liability (cont'd)

- B) (cont'd)
 - 2) license preparation, processing, and related fees;
 - 3) tariff preparation, processing, and related fees;
 - 4) cost of removal and restoration, where appropriate; and
 - 5) any other identifiable costs related to the specially constructed or rearranged facilities.
- C) The applicable termination liability method for calculating the unpaid balance of a term obligation. The amount of such charge is obtained by multiplying the sum of the amounts determined as set forth in Section 9.1.3(B) preceding by a factor related to the unexpired period of liability and the discount rate for return and contingencies. The amount determined in Section 9.1.3(B) preceding shall be adjusted to reflect the redetermined estimate net salvage, including any reuse of the facilities provided. This product is adjusted to reflect applicable taxes.

9.2 Individual Case Basis (ICB) Arrangements

Arrangements will be developed on a case-by-case basis in response to a bona fide special request from a Customer or prospective Customer to develop a competitive bid for a service not generally available under this tariff. ICB rates will be offered to the Customer in writing and on a nondiscriminatory basis. Contracts resulting from a special request will be submitted for approval and filed with the PUCO

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

9.0 <u>SPECIAL ARRANGEMENTS</u> (cont'd)

9.3 Special Promotions

The Carrier may from time to time engage in special promotional trial service offerings of limited duration (not to exceed ninety days on a per customer basis for non-optional, recurring charges) designed to attract new subscribers or to increase subscriber awareness of a particular tariff offering. Requests for promotional offerings will be presented to the Commission for its review in accordance with rules and regulations established by the Commission, and will be included in the Carrier's tariff as an addendum to the Carrier's price lists.

9.4 Discounts

The Company may, from time to time as reflected in the price list, offer discounts based on monthly volume (or, when appropriate, "monthly revenue commitment" and/or "time of day" may also be included in the tariff).

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

10.2 <u>Standard Business Line</u> (see Section 6.3 of this tariff for Price Ranges).
 10.2.1 <u>Single Line and Multiple Line Business Customers</u> (see Section 6.3.1 of this tariff for Price Ranges).

Monthly Recurring Line Charge, Ameritech Serving Area (Touch Tone Included):

Term Plan	Monthly Recurring Charge
Monthly	\$19.90
12 Month	\$19.50
24 Month	\$19.25
36 Month	\$18.80
48 Month*	
60 Month*	

*48 and 60 month customers with service contracts prior to April 1, 1999, will not experience changes in their service rates or conditions for the duration of their contract. Term Plans reflected above, at current rates and conditions will occur if the customer chooses to renew their contract after April 1, 1999. (N) (N)

Monthly Recurring Line Charge, Cincinnati Bell Serving Area (Touch Tone Included):

Term Plan	Monthly Recurring Charge
Monthly	\$38.40
12 Month	\$34.70
24 Month	\$33.30
36 Month	\$32.40

(D) (D)

(D)

Other Standard Business Line Plus Charges and Credits:

Monthly Recurring Charges

911 Emergency Charge		
Ameritech Serving Area	\$0.12 Per Circuit	
Cincinnati Serving Area	\$0.24 Per Circuit	
-Does not apply to DID or inbound only ed	juipped circuits	
End-User Common Line (EUCL) **	\$3.50 Per Circuit/Single Line	(T)
-Also known as FCC Charge	\$5.32 Per Circuit/Multi Line	(N)
TDD Surcharge	\$0.00	~ /
5		(\mathbf{T})
		(1)

**(Ameritech Only) The Company may waive these charges by publishing a Tariff Price List reflecting no charge.

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 PUCO Tariff No. 2 3rd Revised Page 133

Cancels 2nd Revised Page 133

- **10.2** <u>Standard Business Line</u> (cont'd) (see Section 6.3 of this tariff for Price Ranges).
 - **10.2.1** <u>Single Line and Multiple Line Business Customers</u> (cont'd) *(see Section 6.3.1 of this tariff for Price Ranges).* (T)

Monthly Recurring Line Cha	arge Volume Discounts: None	
Non Recurring Installation (Charge (Ameritech Serving Area):	(M)
Per Order:	\$45.00 1 st Line \$20.00 Aditional Lines	
Local Per Call Charges: Ameritech Serving Area	\$0.060	(M) (T)
Non Recurring Installation (Charge (Cincinnati Bell Serving Area):	(N)
Per Order:	\$45.00 1 st Line \$20.00 each additional	
		(N)

IntraLATA IntraState Toll Charges:

See Section 6.9.1 for Price Ranges, See Section 10.8.1 for Price List Rates.

ICG Toll Rounding Policy:

Call Minimum	18 seconds
Call Rounding	6 seconds

* Local Calling Service Areas are listed in Section 4.1 of this tariff.

Issued: December 11, 1998

Effective: December 11, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

10.2 Standard Business Line (cont'd) (see Section 6.3 of this tariff for Price Ranges).

*Information previously located on this page is now located on Page 134.

Standard Business Line Early Termination Penalty Charge:

If the Customer terminates their Standard Business Line Term Plan prior to its completion, the Customer will be responsible for reimbursing ICG for the difference between the Term Plan monthly recurring charge and the Month-to-Month monthly recurring charge for all completed or partial months from the beginning of the agreement period for each of the Customers' Standard Business Lines covered under the Term Plan. In addition, the Customer must reimburse ICG for all installation, wiring or other initial fees that may have been waived at time of initial service.

Issued: December 11, 1998

Effective: December 11, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

Englewood, CO 80112

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)

10.2 <u>Standard Business Line</u> (cont'd) 10.2.3 <u>ICG Standard Business Line Plus</u>

Monthly Recurring Line Charge, Ameritech Serving Area (Touch Tone Included):

Term Plan	Monthly Recurring Charge
Monthly	\$19.90
12 Month	\$19.50
24 Month	\$19.25
36 Month	\$18.80
48 Month*	
60 Month*	

*48 and 60 month customers with service contracts prior to April 1, 1999, will not experience changes in their service rates or conditions for the duration of their contract. Term Plans reflected above, at current rates and conditions will occur if the customer chooses to renew their contract after April 1, 1999.

Monthly Recurring Line Charge, Cincinnati Bell Serving Area (Touch Tone Included):

Monthly \$40.70 12 Month \$39.30 24 Month \$36.00	Term Plan	Monthly Recurring Charge
· · · · · · · · · · · · · · · · · · ·	Monthly	\$40.70
24 Month \$36.00	12 Month	\$39.30
	24 Month	\$36.00
36 Month \$34.65	36 Month	\$34.65

Other Standard Business Line Plus Charges and Credits:

Monthly Recurring Charges

	monthly moduling ondigos	
911 Emergency Charge		
Ameritech Serving Area	\$0.12 Per Circuit	
Cincinnati Serving Area	\$0.24 Per Circuit	0.0
*Does not apply to DID or inbour	nd only equipped circuits	(N)
Ameritech Serving Area EUCL	\$3.50 Per Circuit/Single Line	(N)
-	\$5.32 Per Circuit/Multi Line	
TDD Surcharge	\$0.00	
-		

Issued: March 1, 1999

Effective: April 1, 1999

(D) (D)

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-1650-TP-TRF.

10.2 Standard Business Line (cont'd)

10.2.3 ICG Standard Business Line Plus (cont'd)

Standard Business Line Plus Features Package:

3-Way Calling	N/C
Consultation Hold	N/C
Touch Tone	N/C
Call Transfer	N/C
Hunting	N/C

Standard Business Line Plus Optional Features (Ameritech Serving Area):¹

Call Forward Variable	\$1.50
Call Forward Busy	\$1.50
Call Forward No Answer	\$1.50
Message Waiting	\$1.50

¹ This feature package is available only to customers who have signed service contracts prior to 4/1/99. Please refer to Section 10.9 for Price List rates on Custom Calling Features for contracts signed after 4/1/99.

Issued: November 1, 1999

Effective: December 1, 1999

(C)

(C)

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-1650-TP-TRF.

Single and Multiple Analog PBX Trunk (see Section 6.4 of this tariff for Price Ranges). 10.3

On-Switch PBX DS-0 Trunk Volume Discounts (Ameritech Serving Area): (T) Volume discounts are based on the number of lines and term plan as described below.

Monthly Recurring PBX DS-0 Trunk Charges, Ameritech Serving Area (Includes Touch Tone):

Trunks	Month to Month	12 Month	24 Month	36 Month	48 Month	60 Month
1-47	\$25.37	\$25.14	\$24.75	\$24.18	\$23.84	\$23.45
48-96	\$24.66	\$24.44	\$24.06	\$23.43	\$23.18	\$22.80
97-199	\$23.96	\$23.75	\$23.38	\$22.76	\$22.52	\$22.15
200-293	\$23.25	\$23.05	\$22.70	\$22.10	\$21.86	\$21.50
294-387	\$22.55	\$22.36	\$22.01	\$21.43	\$21.20	\$20.85
288-579	\$21.84	\$21.66	\$21.33	\$20.77	\$20.54	\$20.21
580+	\$21.14	\$20.97	\$20.64	\$20.10	\$19.88	\$19.56

Applies to inbound, outbound, DID, DOD, 2-Way DID circuits.

Monthly Recurring PBX DS-0 Trunk Charges, Cincinnati Bell Serving Area (Includes Touch Tone) (N)

Term Plan	Price
Month to Month	\$ 59.00
12 Month	\$ 57.00
24 Month	\$ 56.00
36 Month	\$ 55.00
48 Month	\$ 53.00
60 Month	\$ 51.00

Other Trunk Service Charges and Credits:				
	Monthly Recurring Charges			
911 Emergency Charge				
Ameritech Serving Area	\$0.12 Per Circuit			
Cincinnati Serving Area	\$0.24 Per Circuit			
*Does not apply to DID or inbound only equipped circuits				
TDD Surcharge	\$0.00			

Issued: December 11, 1998

Effective: December 11, 1998

(T)

(N) (T)

(D)

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

10.3 Single and Multiple Analog PBX Trunk (cont'd) (see Section 6.4 of this tariff for Price Ranges).

Monthly Recurring Line Char	ge Volume Discounts: None			
Non Recurring Installation Charge, Ameritech Serving Area:		(T)		
Per Order:	\$45.00 1st Line \$20.00 Additional Lines			
Local Per Call Charge:	\$0.060/call			
Non Recurring Installation Charge, Cincinnati Bell Serving Area:		(N)		
Per Order	\$60.00	 (N)		
IntraLATA IntraState Toll Charges: See Section 6.9.1 for Price Ranges, See Section 10.8.1 for Price List Rates.				
ICG Toll Rounding Policy: Call Minimum	18 seconds			

* Local Calling Service Areas are listed in Section 4.1 of this tariff.

PBX DS-0 Trunk Early Termination Penalty Charge:

Call Rounding

If the Customer terminates their PBX DS-0 Trunk Term Plan prior to its completion, the Customer will be responsible for reimbursing ICG for the difference between the Term Plan monthly recurring charge and the Month-to-Month monthly recurring charge for all completed or partial months from the beginning of the agreement period for each of the Customers' PBX DS-0 Trunks covered under the Term Plan. In addition, the Customer must reimburse ICG for all installation, wiring or other initial fees that may have been waived at time of initial service.

6 seconds

Issued: December 11, 1998

Effective: December 11, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

10.4 <u>Direct Inward Dial Service (DID)</u> (see Section 6.5 of this tariff for Price Ranges).

Monthly Recurring Charges, Ameritech Serving Area:

Per DID Equipment Line/ Trunk Charge	\$ 5.00
1 st Block of 20 DID Numbers	\$ 2.75
Additional Blocks of 20 DID Numbers	\$ 2.75

Monthly Recurring Charges, Cincinnati Bell Service Area:

Per DID Trunk	\$ 5.00
First Block of 20 DID Numbers	\$ 3.00
Additional Blocks of 20 Numbers	\$ 2.00
First Block of 20 Numbers over 100	\$ 1.50

Non-Recurring DID Installation Charges, Ameritech Serving Area:

Per DID Equipped Line/ Trunk Charge	\$220.00
1 st Block of 20 DID Numbers	\$100.00
Additional Blocks of 20 DID Numbers	\$ 35.00

Non-Recurring DID Installation Charges, Cincinnati Bell Serving Area:

Per DID Trunk	\$ 20.00
First Block 20 DID Numbers	\$100.00
Additional Blocks of 20 DID Numbers	\$ 20.00
First Block of 20 DID Numbers	
over 100	\$ 20.00

*DID Equipped Line/Trunk MRCs are in addition to the Standard Business Line MRCs, the PBX DS-0 Trunk MRCs, or the Digital DS-1 Trunk active channel MRCs. In addition, the Customer will be charged for the number of DID Number Blocks (20 numbers per block) regardless of the number of DID numbers utilized out of the available 20 numbers.

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 (R)

(R)

10.4 <u>Direct Inward Dial Service (DID)</u> (cont'd) (see Section 6.5 of this tariff for Price Ranges).

10.4.1 Direct Outward Dial (DOD) and 2-Way DID Service, Ameritech Serving Area	(T)
(see Section 6.5.1 of this tariff for Price Ranges).	

Monthly Recurring Charges:

Per DOD Equipped Line Trunk Charge	\$ 0.00
Per 2-Way DID Equipped Line/Trunk Charge	\$20.00
1 st Block of 20 DID Numbers	\$ 2.75
Additional Blocks of 20 DID Numbers	\$ 2.75

Non-Recurring DOD and 2-Way DID Installation Charges:

Per DOD Equipped Line/Trunk Charge	\$	0.00
Per 2-Way DID Equipped Line/Trunk Charge	\$22	20.00
1 st Block of 20 DID Numbers	\$1(0.00
Additional Blocks of 20 DID Numbers	\$ 3	35.00

*DOD and 2-Way DID Equipped Line/Trunk MRCs are in addition to the Standard Business Line MRCs, the PBX DS-0 Trunk MRCs, or the Digital DS-1 Trunk active channel MRCs. In addition, the Customer will be charged for the number of DID Number Blocks (20 numbers per block) regardless of the number of DID #s utilized out of the available DID 20 number blocks.

Issued: December 11, 1998

Effective: December 11, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)

10.5 <u>Digital Voice Grade DS-1 Trunk Service</u> (see Section 6.6 of this tariff for Price Ranges).

Digital Voice Grade DS-1 Trunk Service provides line connections from an end user's Private Branch Exchange ("PBX"), Key System, or other telephonic device to the Switch Port. Digital Voice Grade DS-1 Trunk Service is available as a DS1 (1.544 Mbps). Service on a Digital Trunk is available on a measured rate basis for business customers only. A minimum order of 12 voice grade connections (active DSO trunks) is required. Charges for Local Calling Service and Message Telecommunications Service are assessed on a measured rate basis and are additional to charges for Digital Voice Grade DS-1 Trunk Service. These charges only apply to ICG Customers. *(see Sections 6.4 and 6.9 for Price Ranges, see Sections 10.3 and 10.8 for Price Lists).*

The Customer may opt to utilize Digital Voice Grade DS-1 Trunk Service for outgoing calls only, or for outgoing and incoming calls. In the latter case, direct inward dial ("DID") numbers must be purchased from the Company.

. . .

10.5.1.1	Digital Voice DS-1 Trunk Service, Ameritech Serving Area (see Section 6.6.1.1 of this tariff for Price Ranges).		(T)	
	Basic Digital Trunk Option		(N)	
	The basic digital trunk option must include a Ds-1 loop connection, unless customer is collocated with an ICG facility.			
	Monthly Recurring DS-1 L	oop Connection Charges:	(N) (T)	
	<u>Term Plan</u>	Monthly Recurring Charge	i	
	Monthly	\$150.00	Ì	
	12 Month	\$150.00		
	24 Month	\$150.00	I	
	36 Month	\$150.00		
	48 Month	\$150.00		
	60 Month	\$150.00	1	
			(T)	
	Non Recurring Charge		(M)	
	\$500.00		 (M)	
* Material on this page was previously located on page 142.				

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

	verness Drive W		3 rd Revised Page 14	2
	wood, CO 80112		Cancels 2 nd Revised Page 14	
¥			TWORK SERVICES - RETAIL (PRICE LIST)	
10.0				
10.5		rade DS-1 Trunk Service (of this tariff for Price Range		
	(see section 0.0	of this tariff for I rice Range	-5).	
	10.5.1.1 Digits	al Voice Grade DS-1 Trunk	<u>x Service (24 Channels)</u> (cont'd)	
		ection 6.6.1 of this tariff for		
	(5 555	3 /	
	DS0 charge - a	pplies to 2-way, DID or DO	D, Ameritech Service Area	(T)
	Term Plan	Monthly RecurringCh	harge (Per Active DS0)	(TT)
		Service Orders signed	Service orders signed	(T)
		before 12/1/99	on or after 12/1/99	(N)
	Monthly	\$20.46	\$23.00	(I)
	12 Month	\$19.96	\$21.00	
	24 Month	N/A	\$18.50	İ
	36 Month	N/A	\$17.00	1
	48 Month	N/A	\$16.00	1
	60 Month	N/A	\$15.00	1
				(I) (D)
	Hunting Charge	s do not apply		(12)
	Non Recurring	Installation Charge, Amerite	ch Serving Area:	
	Per Orde			(M)
		per order	\$ 45.00	(M)
		al DS0 per order	\$ 20.00	(M)(T)
				(M)(T) (M)(T)
	OtherTrunk Se	ervice Charges and Credits	:	
		8	Monthly Recurring Charges	
	011 Em	ergency Charge		
		ch Serving Area	\$0.12 Per Circuit	
	Cincinne	ati Serving Area	\$0.24 Per Circuit	
		s not apply to DID or inboun		(T)
	Does	s not apply to DID of mooun	a only equipped encurs	(1)
	TDD Su	rcharge	\$0.00	
	Monthly Recu	rring DS-1 Trunk Volume	Discounts: None	
	* Material previ	iously on this page is now or	n page 141.	

Issued: November 1, 1999

ICG Telecom Group, Inc.

Effective: December 1, 1999

PUCO Tariff No. 2

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

(D)

(D)

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)

10.5 Digital Voice Grade DS-1 Trunk Service (cont'd) (see Section 6.6 of this tariff for Price Ranges).

10.5.1.1 <u>Digital Voice DS-1 Trunk Service, Ameritech Serving Area (24 Channels)</u> (cont'd) (see Section 6.6.1 of this tariff for Price Ranges).

Local Per Call Charge: \$0.060/call

IntraLATA IntraState Toll Charges: See Section 6.9.1 for Price Ranges, See Section 10.8.1 for Price List Rates.

ICG Toll Rounding Policy:

Call Minimum Call Rounding 18 seconds 6 seconds

* Local Calling Service Areas are listed in Section 4.1 of this tariff.

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

10.5 Digital Voice Grade DS-1 Trunk Service (cont'd)

(see Section 6.6 of this tariff for Price Ranges).

A) Basic Digital Trunk Option (Cincinnati Bell Serving Area)

The basic Digital trunk option must include a DS-1 loop connection, unless customer is collocated with an ICG facility.

1) DS-1 Loop Connection

Monthly Recurring Charge \$150.00

Non-Recurring Charge \$500.00

2) DS0 Charge - Applies to 2-way, DID or DOD

Non-Recurring Charge (Per Active DS0):1st DS0 per order\$45.00£ach additional DS0 per order\$20.00

Monthly Recurring Charge (Per Active DS0):

Month-to-Month	\$35.00
12 Month	\$34.00
24 Month	\$31.00
36 Month	\$29.00
48 Month	\$28.50
60 Month	\$27.00

B) Fully Configured Digital Trunk Option - Includes DS-1 Loop and 24 Active DS0s (Cincinnati Bell Serving Area)

Non-Recurring Charges

\$1000.00

Monthly Recurring Charge (Per DS-1)

Month-to-Month	\$978.00
12 Month	\$944.00
24 Month	\$766.00
36 Month	\$696.00
48 Month	\$662.00
60 Month	\$628.00

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

10.5 <u>Digital Voice Grade DS-1 Trunk Service</u> (cont'd) (see Section 6.6 of this tariff for Price Ranges).

(D)

(D)

10.5.1.2 Three Way Calling/Call Transfer Feature, Ameritech and Cincinnati Bell Serving Areas

Available on digital trunks only, this feature provides functionality allowing a station user to transfer any established call to another station outside the PBX or business group without the assistance of the attendant. The customer may also add on another party for a three way call. Customers utilizing this service will be billed for any toll charges applicable to the call.

Nonrecurring ChargesMonthly Recurring Charges\$50.00 per trunk group\$3.00 per trunk

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

10.5 Digital Voice Grade DS-1 Trunk Service (cont'd)

(see Section 6.6 of this tariff for Price Ranges).

(D) (N)

(N)

(D)

10.5.2 <u>Fully Configured Digital Trunk Option - includes DS-1 Loop and Active 24</u> <u>DS0s Ameritech Serving Area</u>

Non-recurring charges

\$1000.00

Monthly Recurring Charge (per DS-1)

Month-to-Month	\$690.00
12 Month	\$596.00
24 Month	\$490.00
36 Month	\$420.00
48 Month	\$398.00
60 Month	\$388.00

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

10.5 <u>Digital Voice Grade DS-1 Trunk Service</u> (cont'd) (see Section 6.6 of this tariff for Price Ranges).

10.5.2.1 ISP Monthly Recurring Active DS0 charge - Applies to 2-way, DID, DOD. (T)

(D) |

<u>Term Plan</u>	Monthly Recurring Charge**
Monthly	\$15.93 Per Active DS-1 Channel
12 Month	\$15.82 Per Active DS-1 Channel
24 Month	\$15.71 Per Active DS-1 Channel
36 Month	\$14.96 Per Active DS-1 Channel
48 Month	\$14.71 Per Active DS-1 Channel
60 Month	\$14.21 Per Active DS-1 Channel

*Applies to all active inbound, outbound, DID, DOD, 2-Way DID DS-1 channels *Hunting Charges do not apply

**Pricing applies for service orders signed before 12/1/99. Pricing options available after 12/1/99 are found in 10.5.1. of this Tariff.

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

10.5 <u>Digital Voice Grade DS-1 Trunk Service</u> (cont'd) (see Section 6.6 of this tariff for Price Ranges).

10.5.2.1 ISP Monthly Recurring Active DS0 charge - Applies to 2-way, DID, DOD

(D)

(D)

(T)

Local Per Call Charge:

\$0.060/call

IntraLATA IntraState Toll Charges: See Section 6.9.1 for Price Ranges, See Section 10.8.1 for Price List Rates.

ICG Toll Rounding Policy: Call Minimum Call Rounding

18 seconds 6 seconds

* Local Calling Service Areas are listed in Section 4.1 of this tariff.

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

10.5 <u>Digital Voice Grade DS-1 Trunk Service</u> (cont'd) (see Section 6.6 of this tariff for Price Ranges).

10.5.2.2 <u>Internet Service Provider (ISP) DS-1 Trunk Service, Ameritech Service Area</u> (T) (34 Channels) (see Section 6.6.2 of this tariff for Price Ranges).

Digital DS-1 Trunk and Active Channel Early Termination Penalty Charge:

If the Customer terminates their Digital Data DS-1 Trunk and Active Channel Term Plan prior to its completion, the Customer will be responsible for reimbursing ICG for the difference between the Term Plan monthly recurring charge and the Month-to-Month monthly recurring charge for all completed or partial months from the beginning of the agreement period for each of the Digital Data DS-1 Trunk and Active Channel covered under the Term Plan. In addition, the Customer must reimburse ICG for all installation, wiring or other initial fees that may have been waived at time of initial service.

Issued: November 1, 1999

Effective: December 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

10.5 Digital Voice Grade DS-1 Trunk Service (cont'd) (see Section 6.6 of this tariff for Price Ranges).

10.5.3 <u>Integrated Services Digital Network (ISDN) Primary Rate Interface (PRI)</u> (see Section 6.6.3 of this tariff for Price Ranges).

Monthly Recurring Charge:

<u>Term Plan</u>	<u>1st PRI</u>	2 nd PRI	<u>3rd PRI & Higher</u>
Monthly	\$350.00	\$250.00	\$200.00
12 Month	\$340.00	\$240.00	\$190.00
24 Month	\$330.00	\$230.00	\$180.00
36 Month	\$320.00	\$220.00	\$170.00
48 Month	\$315.00	\$215.00	\$165.00
60 Month	\$310.00	\$210.00	\$160.00

• ISDN PRI MRCs are in addition to the Digital T1 Facility MRCs (See Section 10.5 of this tariff for Digital T1 Monthly Recurring Charges).

ISDN PRI Installation Charge:

\$1,750.00/T1 Facility

Term Plan Price Standard Features Include:

- 23B+D
- 24B; Requires purchase of one 23B+D
- "D" Channel Control of Multiple ISDN Primes
- 64 Clear Channel Capability
- Direct Inward Dialing
- Dedicated Trunk Groups

Digital DS-1 Trunk is an additional fee Refer to Digital DS-1 Trunk Pricing for Dialtone

DID Trunk Termination:

DID channel termination fees are eliminated do to the non-channelization of the PRI trunk. A monthly fee for each telephone number will apply - refer to the DID pricing portion of this tariff. *(See Section 6.5 for Price Ranges, See Section 10.4 for Price List Rates)*

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

10.5 <u>Digital Voice Grade DS-1 Trunk Service</u> (cont'd) (see Section 6.6 of this tariff for Price Ranges).

10.5.3 <u>Integrated Services Digital Network (ISDN) Primary Rate Interface (PRI)</u> (cont'd) (see Section 6.6.3 of this tariff for Price Ranges).

ISDN PRI Optional Feature:

	<u>Term</u>	Monthly Recurring Charge
23B with "D"		
Channel Backup	1 Month Plan	\$110.00
	12 Month Plan	\$108.00
	24 Month Plan	\$106.00
	36 Month Plan	\$104.00
	48 Month Plan	\$102.00
	60 Month Plan	\$100.00

(The D Channel backup is provisioned separately from the 23B (which is provisioned as a 23B+D)

Additions, Deletions, Rearrangements &	Installation
Changes of one or more Trunks to existing	
Trunk Groups:	
Per Interface, Occasion or Trunk Group	\$50.00

Issued: May 24, 1998

Effective: May 24, 1998

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

10.8 Message Telecommunications Service (see Section 6.9 of this tariff for Price Ranges).

10.8.1 Switched Outbound Service

10.8.1.1 IntraLATA Switched Outbound Service

IntraLATA Switched Inbound Service is billed in six (6) second increments, with an six (6) second call minimum. After the initial 6 second minimum, each fractional call is rounded up to the next one-tenth minute. Customers will receive the prices below 24 hours a day, 7 days a week.

Ameritech Serving Area

<u>Usage</u>	age <u>Peak</u> <u>Off-Peak</u>	
Analog Service Digital Service	\$0.079 \$0.060	\$0.060 \$0.055
Peak Off-Peak	Monday - Friday Monday - Friday Saturday, Sunday Holidays	8:00am - 4:59pm 5:00pm - 7:59am All Day All Day

All times are Central Standard or Central Daylight Savings Time.

Cincinnati Bell Serving Area

All Services, 24 Hours Per	\$0.079
Day, 7 Days Per Week	

10.8.1.2 InterLATA/Intrastate Switched Outbound Service

InterLATA/Intrastate Switched Outbound Service is billed in six (6) second increments, with an six (6) second call minimum. After the initial 6 second minimum, each fractional call is rounded up to the next one-tenth minute. Customers will receive the prices below 24 hours a day, 7 days a week.

<u>Usage</u>

Outbound Usage	\$0.079 per minute
On-Switch Customer Discount (applicable on per minute usage charges)	10%

Issued: September 29, 2000

Effective: October 1, 2000

(R)

(R)

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

10.8 Message Telecommunications Service (cont'd)

10.8.2 Switched Inbound Service (ICG Toll Free)

10.8.2.1 IntraLATA Switched Inbound Service

IntraLATA Switched Inbound Service is billed in six (6) second increments, with a six (6) second call minimum. After the initial 6 second minimum, each fractional call is rounded up to the next one-tenth minute. Customers will receive the prices below 24 hours a day, 7 days a week.

<u>Usage</u>	
Inbound Usage	\$0.079 per minute
On-Switch Customer Discount	10%
(applicable on per minute usage charges)	
Per -Call Payphone Service Charge:	\$0.250

10.8.2.2 InterLATA/Intrastate Switched Inbound Service

InterLATA/Intrastate Switched Inbound Service is billed in six (6) second increments, with a six (6) second call minimum. After the initial 6 second minimum, each fractional call is rounded up to the next one-tenth minute. Customers will receive the prices below 24 hours a day, 7 days a week.

Usage	
Inbound Usage	\$0.079 per minute
On-Switch Customer Discount	10%
(applicable on per minute usage charges)	
Per-Call Payphone Service Charge:	\$0.250
• •	

108.2.3 MTS Services - Plan B

Plan B covers all instate long distance services (IntraLATA/Intrastate, InterLATA/Intrastate, outbound and inbound calls) with one rate which is applied to all usage. In order to qualify for Plan B service, customers must sign a minimum one year term agreement for ICG long distance service, and select ICG as their primary interexchange carrier for both InterLATA and IntraLATA long distance. All service is billed in six (6) second increments with a six (6) second call minimum.

Usage\$0.050Per-Call Payphone Service Charge (inbound only)\$0.250

Issued: September 29, 2000

Effective: October 1, 2000

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

10.9 Business Custom Calling Features, Ameritech Serving Area (see Section 6.10 of this tariff for Price Ranges).

Feature Description	Monthly Recurr	ring Charges	Non Recurring Charges
Hunting	\$3	3.00	\$2.00
Call Forwarding	\$3	3.50	\$7.40
Call Forwarding Busy	\$3	3.50	\$7.40
Call Forwarding No A	nswer \$3	3.50	\$7.40
Call Waiting		5.50	\$7.40
	11	2.50	\$7.40
3-Way Conference Ca		3.50	\$7.40
Speed Call (8)		3.50	\$7.40
Speed Call (30)		3.50	\$7.40
Caller ID(w/o name)	\$1	7.50	\$7.40
Caller ID(with name)	\$8	8.50	\$7.40
Retained # (DNCF) wi	ith 4 Paths \$4	4.00	\$50.00
Each Additional		1.00	\$0.00
Vanity Number	\$(0.00	\$32.40
Touch Tone	\$0	0.00	\$0.00
911 Location Manager	•		
- Per DS0 for Dig		8.00	\$50.00
- Per PRI		8.00	\$50.00

Issued: July 12, 2002

Effective: August 12, 2002

(N) (N) (N)

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

Business Custom Calling Features, Cincinnati Bell Serving Area:

Feature Description	Monthly Recurring Charges
Call Waiting	\$4.50
3-Way Calling	\$2.40
Call Forward (Variable)	\$2.40
Call Forward (Busy)	\$0.80
Call Forward (No Answer)	\$0.80
Speed Call - 8	\$2.70
Speed Call - 30	\$2.70
Caller ID (Without Name)	\$7.25
Caller ID (With Name)	\$8.50
Hunting	\$6.50
*DCNF - Listed #'s (up to 4 paths)	\$0.00
*DCNF - Unlisted #'s (Up to 4 paths)	\$14.50
*DCNF - Each additional path	\$1.00
*Touch Tone	\$0.00
911 Location Manager	
- Per DS0 for Digital Trunks	\$58.00
- Per PRI	\$58.00

(N) | |

(N)

Non Recurring Charges

Feature Installation,	\$5.00	
per order, per location 911 Location Manager	\$50.00	(N)
*Non recurring charges do not apply.		(1)

Issued: July 12, 2002

Effective: August 12, 2002

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

10.9 Business Custom Calling Features, Ameritech Serving Area (cont'd) (see Section 6.10 of

this tariff for Price Ranges).

The following Custom Calling Features are available for Volume Discount:*

- Call Forwarding
- Call Forwarding Busy
- Call Forwarding No Answer
- Call Waiting
- 3-Way Conference Calling
- Speed Call (8)
 Speed Call (30)

Spool Carr (50)	Monthly Recurring Charge* Discounts
2 Custom Calling Features per Line	20%
3 Custom Calling Features per Line	30%
4 Custom Calling Features per Line	35%
5 or more Custom Calling Features per Line	40%

Business Custom Calling Features, Cincinnati Bell Serving Area*

The Cincinnati Bell Serving Area Custom Calling Features listed on the preceding page of this tariff can be combined for a quantity discount. The quantity discount is on a per line, per month basis and the total discount amount for all lines will be deducted from the total monthly recurring charges per account for custom calling features.

	Monthly Recurring Discount
2 Custom Calling Features per Line	\$ 0.75
3 Custom Calling Features per Line	\$ 1.00
4 Custom Calling Features per Line	\$ 1.75
5 or more Custom Calling Features per Line	\$ 2.95

*Business custom calling feature discount is not available for service orders signed after November (T) 5, 1999.

Issued: October 4, 1999

Effective: November 5, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

10.10 <u>900/976Blocking/Unblocking</u> (see Section 6.11 of this tariff for Price Ranges). Customers are given the option of per call blocking at no charge on a per line blocking at a rate listed below

Ameritech Serving Area

Per Call Blocking	N/C
Per Line Blocking	N/C

Cincinnati Bell Serving Area

Per Line Blocking N/C

10.11 Miscellaneous

10.11.1 <u>Maintenance Visit Charges</u> (see Section 6.12 of this tariff for Price Ranges). (Ameritech and Cincinnati Bell Serving Areas)

	Duration of time, per technician	<u>First Hour</u>	Each Addition minutes	<u>nal 15</u>	
	Monday-Friday (8am-5pm) Monday-Friday (5pm-8am) & Saturday Sunday All ICG Recognized Holidays Trouble Isolation Charge (Flat Rate)	\$ 175.00 \$ 184.00 \$ 184.00 \$ 70.00	\$ 131.25 \$ 44.00 \$ 46.00 \$ 46.00	\$ 33.00	
10.11.2	<u>Order Processing Charge</u> Cincinnati Bell Serving Area				(M)
	Non Recurring Charges Per Service Order, per location Per PIC Code Change	\$25.00 \$ 5.00			 (M) (R)(M) (M)
	Ameritech Serving Area <u>Non Recurring Charges</u> Per Service Order, per location Per PIC Code Change	\$20.00 \$ 5.00			 (M) (R)(M) (M)

*Material on this page was previously located on page 157.

Issued: February 19, 2001

Effective: March 1, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-249-TP-ATA

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 (T)

(T)

ICG Telecom Group, Inc.PUCO Tariff No. 2161 Inverness Drive West4th Revised Page 156Englewood, CO 80112Cancels 3td Revised Page 15610.0CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)10.12Directory Listings (see Section 6.13 of this tariff for Price Ranges).					
Ameritech Serving Area Standard Residential/Business Lis Additional Residential Listing Additional Business Directory Lis Semi Public Listing (# excluded but included in DA Private Non-Listing (# excluded and DA)	sting sting l from listings L)	Monthly Recurring Charge \$0.00/Line \$0.00/Line \$3.00/Line \$1.50/Line \$1.10/Line	(T)		
<u>Cincinnati Bell Serving Area</u> Primary Listing Regular Additional Listing Alternate Listing Secretarial Listing Private Listing Foreign Listing Extra Line Matter ¹ Non-Address	<u>Monthly Recurring Char</u> \$0.00 \$3.00 \$1.47 \$1.47 \$1.96 \$3.00 \$1.47	rge <u>Non Recurring Charge</u> *\$12.37 \$12.37 \$12.37 \$12.37 \$12.37 *\$12.37 \$12.37 \$12.37 *\$12.37 *\$12.37 *\$12.37	(N) 		

*The non recurring charge applies only when changes are made to the listing after the line has been established. ¹Extra Line Matter is available on a Primary Listing, Regular Additional Listing and Secretarial Listing.

10.13 <u>Calling Card Services</u> (see Section 6.14 of this tariff for Price Ranges).

Charge per minute

\$0.276 per minute (within continental US)

ICG's Calling Card Service is offered to all ICG local exchange and long distance customers or as a standalone service. Service is offered twenty-four (24) hours a day, seven (7) days a week to all valid terminating locations. Intrastate service is sold as an add-on to interstate service. Access to ICG's Calling Card Service is via a toll free number. The Customer must input a valid Authorization Code in addition to the destination number with area code.

Calling Card calls are billed in six (6) second increments with a thirty (30) second call minimum. After the initial thirty (30) second minimum, each fractional call is rounded up to the nearest one-tenth of a minute. Fractional cents will be rounded to the nearest cent.

Usage of the Calling Card affirms that the Customer agrees to the terms and conditions set forth by this tariff and supporting material sent to the Customer together with the Calling Card.

Issued: March 1, 1999

Effective: April 1, 1999

(N)

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 98-1650-TP-ATA.

(M)

(M)

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)

10.14 Bad Check Charge

\$25.00

*Material previously located on this page is now located on page 155.

Issued: February 19, 2001

Effective: March 1, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-249-TP-ATA

10.15 Digital Access Service

Term / Rate Element	Monthly	1 Year	2 Year	3 Үеаг	4 Year	5 Year	NRC*
PLAN A Central Office Port Option 1***	\$460.00	\$440.00	\$400.00	\$375.00	\$375.00	\$370.00	\$950.00
PLAN A Central Office Port Option 2***	\$414.00	\$396.00	\$360.00	\$303.00	\$356.25	\$351.50	\$950.00
PLAN A Central Office Port Option 3***	\$579.60	\$554.40	\$504.00	\$393.75	\$393.75	\$388.50	\$950.00
PLAN A Digital Access Loop**	\$150.00	\$150.00	\$150.00	\$150.00	\$150.00	\$150.00	\$500.00
PLAN B Central Office Port Option 1	\$376.00	\$364.50	\$336.00	\$303.00	\$265.00	\$253.50	\$950.00
PLAN B Central Office Port Option 2	\$338.00	\$328.00	\$302.00	\$273.00	\$252.00	\$240.00	\$950.00
PLAN B Central Office Port Option 3	\$418.50	\$ 405.00	\$373.50	\$337.50	\$278.00	\$266.00	\$ 950.00
PLAN B Digital Access Loop	\$258.00	\$257.00	\$266.00	\$240.00	\$270.00	\$272.00	\$500.00
- DAS Subgroup (PLAN B only) - Caller Identification - Hunting Within a DAS Subgroup/Arrangement	no charge	no charge	no charge	no charge	no charge	no charge	no charge
Called Number	\$0.00	\$0.00	\$0 .00	\$0.00	\$0.00	\$0.00	\$0.00
Overflow Among DAS Arrangements	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$150.00
Number Blocks (per block of 20 numbers)	\$2.75	\$2.75	\$2.75	\$2.75	\$2.75	\$2.75	\$100.00
Additional Trunk Phone Number	\$5.00	\$5.00	\$5.00	\$5.00	\$5.00	\$5.00	\$0.00
Order Processing Charge - Per Service order, per location	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$20.00
Redirected Number Delivery	\$25.00	\$25.00	\$25.00	\$25.00	\$25.00	\$25.00	\$100.00

(T)

*NRC - Non-recurring charge

**Digital Access Loop charge for Plan A will be waived if customer is collocated.

All rates may be pro-rated for a partial month.

*** For customers entering into contracts on or after 10/1/99.

Issued: February 19, 2001

Effective: March 1, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-249-TP-ATA

10.15 Digital Access Service (cont'd)

Rates and Charges (Ameritech Serving Area): (Per DS-1, unless indicated otherwise)

Term / Rate Element	Monthly	1 Year	2 Year	3 Year	4 Year	5 Year	NRC*
Expanded Originating Service - Routing Establishment Charge (per DAS)	\$25.00	\$25.00	\$25.00	\$25.00	\$25.00	\$25.00	\$150.00
Expanded Originating Service IntraLATA transport (Cleveland Only)	\$125.00	\$125.00	\$125.00	\$125.00	N/A	N/A	\$400.00
Expanded Originating Service - InterLATA Transport Charge Route: Akron MSA - Cleveland MSA	\$200.00	\$200.00	\$200.00	\$200.00	\$200.00	\$200.00	\$400.00
Akron MSA - Columbus MSA	\$440.00	\$440.00	\$440.00	\$440,00	\$440.00	\$440.00	\$400.00
Akron MSA - Dayton MSA	\$680.00	\$680.00	\$680.00	\$680.00	\$680.00	\$680.00	\$400.00
Cleveland MSA - Columbus MSA	\$510.00	\$510.00	\$510.00	\$510.00	\$510.00	\$510.00	\$400.00
Cleveland MSA - Dayton MSA	\$720.00	\$720.00	\$720.00	\$720.00	\$720.00	\$720.00	\$400.00
Columbus MSA - Dayton MSA	\$260.00	\$260.00	\$260.00	\$260.00	\$260.00	\$260.00	\$400.00

Term/Rate Element	2 Year	3 Year	4 Year	5 Year
Plan C - Central Office Port Option 1**	\$350.00	\$325.00	\$300.00	\$275.00
Plan C - Digital Access Loop Option 1	5%	7%	10%	12%

(I)

Discounts based on Plan A, Option 1 pricing. 4 and 5 year term discounts based on Plan A, Option 1 pricing. Please refer to Plan A for all applicable NRC's. -

*NRC - Non-recurring charge

All rates may be pro-rated for a partial month.

** Pricing for service contracts signed on or after 10/1/99.

Issued: October 1, 1999

Effective: October 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

10.15 Digital Access Service (cont'd)

Rates and Charges (Ameritech Serving Area):

Term/Rate Element	2 Year	3 Year	4 Year	5 Year	NRC*
Plan C for DS-3 InterLATA Transport Route:					
Akron to Cleveland	\$1,900.00	\$1,900.00	\$1,900.00	\$1,900.00	\$1,000.00
Akron to Columbus	\$2,875.00	\$2,875.00	\$2,875.00	\$2,875.00	\$1,000.00
Akron to Dayton	\$4,400.00	\$4,400.00	\$4,400.00	\$4,400.00	\$1,000.00
Cleveland to Columbus	\$3,300.00	\$3,300.00	\$3,300.00	\$3,300.00	\$1,000.00
Cleveland to Dayton	\$4,650.00	\$4,650.00	\$4,650.00	\$4,650.00	\$1,000.00
Columbus to Dayton	\$1,900.00	\$1,900.00	\$1,900.00	\$1,900.00	\$1,000.00

Term / Rate Element	3 Year	4 Year	5 Year
Plan C-Central Office Port Option 2 (250-499 PRI)	\$400.00	\$400.00	\$400.00
(500+ PRI)	\$350.00	\$350.00	\$350.00

-Option 2 includes pricing for Central Office Port Only.

-All other DAS features and options are at the rates found previously in this section.

-Please refer to Plan A for all applicable NRC's

Note: Information previously on this page is now located on page 159.0.2

Issued: July 2, 1999

Effective: August 2, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: Julia Waysdorf Vice President, Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 (N) |

(N)

10.0 <u>CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)</u> 10.15 <u>Digital Access Service</u> (cont'd)

Term/Rate Element	Monthly	1 Year	2 Year	3 Year	NRC*
Plan D - DS-3 Digital Access Loop	\$1,500.00	\$1,500.00	\$1,500.00	\$1,500.00	\$500.00
Plan D - DS-3 EOS IntraLATA Transport (Cleveland Only)	\$1,250.00	\$1,250.00	\$1,250.00	\$1,250.00	\$400.00
Plan D EOS InterLATA Transport Route:					
Akron to Cleveland	\$2,300.00	\$2,300.00	\$2,300.00	\$2,300.00	\$500.00
Akron to Columbus	\$3,960.00	\$3,960.00	\$3,960.00	\$3,960.00	\$500.00
Akron to Dayton	\$6,120.00	\$6,120.00	\$6,120.00	\$6,120.00	\$500.00
Cleveland to Columbus	\$4,590.00	\$4,590.00	\$4,590.00	\$4,590.00	\$500.00
Cleveland to Dayton	\$6,480.00	\$6,480.00	\$6,480.00	\$6,480.00	\$500.00
Columbus to Dayton	\$2,340.00	\$2,340.00	\$2,340.00	\$2,340.00	\$500.00

Rates and Charges (Ameritech Serving Area): (cont'd)

- DS-3 pricing replaces the Digital Access Loop, EOS IntraLATA transport, and/or EOS InterLATA transport commonly provisioned at the DS-1 level and available in requested markets.

- All other Digital Access Services pricing components still apply. Please refer to pricing in Plan A, Plan B or Plan C for all other applicable charges.

*NRC - Non-Recurring Charge.

All Rates may be pro-rated for a partial month.

Plan E - Central Office Port	Monthly	1 Year	2 Year	3 Year	4 Year	5 Year
25-49 PRI's	\$460.00	\$440.00	\$392.00	\$368.00	\$346.00	\$325.00
50-99 PRI's	\$460.00	\$440.00	\$384.00	\$360.00	\$338.00	\$325.00
100+ PRI's	\$460.00	\$440.00	\$376.00	\$353.00	\$332.00	\$319.00

-Plan E includes Central Office Port only.

-All other DAS features and options are available at the rates found previously in this section.

-Please refer to Plan A for all applicable NRC's.

Effective: January 8, 2000

(M)

(M)

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

10.0 <u>CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)</u> 10.15 <u>Digital Access Service</u> (cont'd)

<u>NRC</u> *	MRC	
	\$408.00**	
eements signed prior to e Port only.		this section.
\$0.00	\$504.00	
	applicable NRC's. eements signed prior to e Port only. ptions are available at to <u>NRC</u>	applicable NRC's. eements signed prior to January 4, 2002 ee Port only. ptions are available at the rates found previously in <u>NRC MRC</u>

Issued: December 4, 2001

Effective: January 4, 2002

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST) 10.15 Digital Access Service (cont'd)

Rates and Charges: (Per DS1, unless indicated otherwise) (Applies to both Ameritech and Cincinnati Bell Serving Areas)

Term/Rate Element	1 Year	2 Year	3 Year	5 Year	Overage Per MOU*
PLAN B High Volume 50,000 MOU*					\$0.0075
MRC**	\$800.00	\$0.00	\$800.00	\$800.00	
NRC***	\$50.00	\$400.00	\$200.00	\$0.00	
LAN B High Volume 100,000 MOU					\$0.0065
MRC**	\$850.00	\$850.00	\$850.00	\$850.00	
NRC***	\$550.00	\$400.00	\$200.00	.0.00	
PLAN B High Volume 250,000 MOU*					\$0.0050
MRC**	\$1,000.00	\$1,000.00	\$1,000.00	\$1,000.00	
NRC***	\$550.00	\$400.00	\$200.00	\$.0.00	
PLAN B High Volume 350,000 MOU*					\$0.0045
MRC**	\$1,150.00	\$1,150.00	\$1,150.00	\$1,150.00	
NRC***	\$550.00	\$400.00	\$200.00	\$.0.00	

- Charges for moves, charges and deletes found in this tariff also apply.

- Plan B High Volume rate includes central office port only; Digital Access Loop Charge waived if customer is collocated.

- Customer must still select Plan B configuration option 1, 2, or 3.

- * MOU Minutes of Use
- ** MRC Monthly Recurring Charge
- *** NRC Non-recurring charge

All rates may be pro-rated for a partial month.

Material on this page previously appeared on page 159.0.3

Issued: December 4, 2001

Effective: January 4, 2002

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 801128 (M)

Original Page 159.0.4

(M)

CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE 10.0 LIST)

Digital Access Service (cont'd) 10.15

Rates and Charges (Cincinnati Bell Serving Area): (Per DS-1, unless indicated otherwise)

Term / Rate Element	Monthly	1 Year	2 Year	3 Year	4 Year	5 Year	NRC*
PLAN A Central Office Port Option 1	\$750.00	\$725.00	\$650.00	\$622.00	\$622.00	\$618.00	\$400.00
PLAN A Central Office Port Option 2	\$713.00	\$689.00	\$618.00	\$591.00	\$590.00	\$587.00	\$400.00
PLAN A Central Office Port Option 3	\$787.00	\$761.00	\$ 682.00	\$653.00	\$658.00	\$648.00	\$400.00
PLAN A Digital Access Loop**	\$150.00	\$150.00	\$150.00	\$150.00	\$150.00	\$150.00	\$200.00
PLAN B Central Office Port Option 1	\$775.00	\$750.00	\$700.00	\$680.00	\$652.00	\$615.00	\$400.00
PLAN B Central Office Port Option 2	\$737.00	\$735.00	\$713.00	\$646.00	\$620.00	\$585.00	\$400.00
PLAN B Central Office Port Option 3	\$813.00	\$787.00	\$714.00	\$684.00	\$665.00	\$645.00	\$400.00
PLAN B Digital Access Loop	\$150.00	\$150.00	\$150.00	\$150.00	\$150.00	\$150.00	\$200.00
- DAS Subgroup (PLAN B only) - Caller Identification - Hunting Within a DAS Subgroup/Arrangement	no charge						
Called Number	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Overflow Among DAS Arrangements	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$150.00
Number Blocks 1 st block of 20 numbers Additional block of 20 numbers 1 st block of 20 numbers over 100	\$ 3.00 \$ 2.00 \$ 1.50	\$100.00 \$ 20.00 \$ 20.00					
Additional Trunk Phone Number	\$5.00	\$5.00	\$5.00	\$5.00	\$5.00	\$5.00	\$0.00
Order Processing Charge - Per Service order, per location	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$25.00
Redirected Number Delivery	\$25.00	\$25.00	\$25.00	\$25.00	\$25.00	\$25,00	\$100.00

*NRC - Non-recurring charge **Digital Access Loop charge for Plan A will be waived if customer is collocated.

All rates may be pro-rated for a partial month.

Issued: February 19, 2001

Effective: March 1, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 01-249-TP-ATA

CURRENT RATES AND CHARGES NETWORK (PRICE LIST) 10.0

10.15 Digital Access Service (cont'd)

Rates and Charges (Cincinnati Bell Serving Area): (Per DS-1, unless indicated otherwise)

Term / Rate Element	Monthly	1 Year	2 Year	3 Year	4 Year	5 Year	NRC*
Expanded Originating Service - Routing Establishment Charge (per DAS)	\$25.00	\$25.00	\$25.00	\$25.00	\$25.00	\$25.00	\$150.00
Expanded Originating Service - InterLATA Transport Charge Route:	TBD						

Term/Rate Element	2 Year	3 Year	4 Year	5 Year
Plan C - Central Office Port Option 1**	\$650.00	\$578.00	\$560.00	\$550.00
Plan C - Digital Access Loop Option 1	5%	7%	10%	12%

Discounts based on Plan A, Option 1 pricing. 4 and 5 year term discounts based on Plan A, Option 1 pricing.

Please refer to Plan A for all applicable NRC's

Term / Rate Element	3 Year	4 Year	5 Year
Plan C-Central Office Port Option 2 (250-499 PRI)	\$650.00	\$650.00	\$650.00
(500+ PRI)	\$600.00	\$600.00	\$600.00

-Option 2 includes pricing for Central Office Port Only.

-All other DAS features and options are at the rates found previously in this section.

-Please refer to Plan A for all applicable NRC's.

** Pricing for service contracts signed on or after 10/1/99.

Issued: October 1, 1999

Effective: October 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112

(R)

CURRENT RATES AND CHARGES NETWORK (PRICE LIST) 10.0

10.15 Digital Access Service (cont'd)

Rates and Charges (Cincinnati Bell Serving Area): (Per DS-1, unless indicated otherwise)

Term / Rate Element	Monthly	1 Year	2 Year	3 Year	NRC*
Plan D - DS-3 Digital Access Loop	\$1,500.00	\$1,500.00	\$1,500.00	\$1,500.00	\$500.00

- DS-3 pricing replaces the Digital Access Loop, EOS IntraLATA transport, and/or EOS InterLATA transport commonly provisioned at the DS-1 level and available in requested markets. - All other Digital Access Services pricing components still apply. Please refer to pricing in Plan A, Plan B or Plan C for all

other applicable charges.

*NRC - Non-recurring charge All rates may be pro-rated for a partial month.

Plan E - Central Office Port	Monthly	1 Year	2 Year	3 Year	4 Year	5 Year
25-49 PRI's	\$750.00	\$725.00	\$637.00	\$616.00	\$611.00	\$597.00
50-99 PRI's	\$750.00	\$725.00	\$625.00	\$610.00	\$605.00	\$592.00
100+ PRI's	\$750.00	\$725.00	\$612.00	\$604.00	\$600.00	\$585.00

-Plan E includes Central Office Port only.

-All other DAS features and options are available at the rates found previously in this section.

-Please refer to Plan A for all applicable NRC's.

Issued: December 4, 2001

Effective: January 4, 2002

10.0 CURRENT RATES AND CHARGES NETWORK (PRICE LIST)

10.15 Digital Access Service (cont'd)

Rates and Charges (Cincinnati Bell Serving Area): (Per DS-1, unless indicated otherwise)_

Plan F - Central Office Port: 5 Year Term Commitment	<u>NRC</u> *	MRC
200,000 + PRIs		\$408.00**

* -Please refer to Plan A For all applicable NRC's.

** -pricing available only for agreements signed prior to January 4, 2002 (N)
 -Plan F includes Central Office Port only.

-All other DAS features and options are available at the rates found previously in this section.

Plan G-Central Office Port: 66 Month Term Commitment	NRC	<u>MRC</u>
100,000 + PRI's		
Months 1-36	\$0.00	\$504.00
Months 37-66	\$0.00	\$288.00

-Plan G includes Central Office Port Only

-All other DAS features and options are available at rates found previously in this section

(Ń)

(M)

(N)

Issued: December 4, 2001

Effective: January 4, 2002

10.0 CURRENT RATES AND CHARGES NETWORK (PRICE LIST)

10.16 Business Line Connect

Ameritech Saving Area Non-Recurring Charges Per Line

\$ 75.00 for first line per order\$ 50.00 each additional line per order

Monthly Recurring Charges

Term Plan	Month to Month	1 Year	2 Year	3 Year
Business Line Connect*	\$38.00	\$36.00	\$34.25	\$32.50
Business Line Connect Plus**	\$45.00	\$43.00	\$40.75	\$38.50

Cincinnati Bell Saving Area Non-Recurring Charges Per Line Per Line

\$ 75.00 for first line per order\$ 50.00 each additional line per order

Monthly Recurring Charges

Term Plan	Month to Month	1 Year	2 Year	3 Year
Business Line Connect*	\$34.00	\$34.00	\$34.00	\$34.00
Business Line Connect Plus**	\$47.90	\$42.00	\$37.50	\$36.00

*Monthly Recurring Charge for Business Line Connect includes the following features: Standard Business Line, Intercom Dialing, Hunting, 3-way Conference Calling, Consultation Hold, Touch Tone, Call Transfer External, Call Forwarding Variable, Call Forwarding Busy and Call Forwarding Don't Answer.

**Monthly Recurring Charge for Business Line Connect Plus includes all features of Business Line Connect and Voice Mail.

Issued: March 1, 1999

Effective: April 1, 1999

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: J. Carl Jackson, Jr. Sr. Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112 (N)

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)

10.17	<u>Dedicate</u> 10.17.1	with an initial minimum billing period of	ed Outbound Service Service is billed in six (6) second increments, f six (6) seconds. After the initial 6 second up to the next one-tenth minute. Customer's	(R)
	10.17.2	with an initial minimum billing period of	Service is billed in six (6) second increments, f six (6) seconds. After the initial 6 second up to the next one-tenth minute. Customer's	(R)
	10.17.3	IntraLATA/Intrastate Dedicated Inbound S with an initial minimum billing period of	ed Inbound Service (Dedicated Toll Free) Service is billed in six (6) second increments, f six (6) seconds. After the initial 6 second up to the next one-tenth minute. Customer's ay, 7 days a week. \$0.079 40% \$0.250	(R)
	10.17.4	InterLATA/Intrastate Dedicated Inbound with an initial minimum billing period of	ed Inbound Service (Dedicated Toll Free) Service is billed in six (6) second increments, f six (6) seconds. After the initial 6 second up to the next one-tenth minute. Customer's ay, 7 days a week. \$0.079 40% \$0.250	(R)

10.17.5Plan A: Dedicated Long Distance Directory Assistance
(NPA) 555-1212\$0.40/call

*Dedicated Customer Discount does not apply.

Issued: January 19, 2000

Effective: January 20, 2000

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST) 10.18 Business Essentials

<u>Component A</u> - includes one standard Business Line Plus line and the following features; hunting, 3 Way calling, call transfer, consultation hold, Caller ID, Call Waiting, Call Forwarding/Variable /Busy/No Answer.

Ameritech Serving Cincinnati Serving		Cincinnati Serving Area
Term	Monthly Recurring	Monthly Recurring Charge
12 Month	\$36.99	\$41.99
24 Month	\$34.99	\$39.99
36 Month	\$32.99	\$37.99

-Customers may elect to decline any of the Custom Calling features included in the Business Essentials Bundle, however this will not impact the monthly recurring rate.

-Additional Custom Calling features are available at the rates listed in this tariff.

<u>Component B</u> - Per minute rate applies to all inbound and outbound IntraLATA/intrastate, InterLATA/intrastate and interstate calls. Volume Discounts apply on a per month, per account basis.

Ameritech Serving Area				
Term	1-2499 MOU	2500-4999 MOU	5000+ MOU	
12 Month	\$0.075	\$0.069	\$0.065	
24 Month	\$0.075	\$0.069	\$0.065	
36 Month	\$0.075	\$0.069	\$0.065	

-Calling Cards and International calls are at the regular tariffed rates.

-This Service is billed in six (6) seconds increments, with an initial minimum billing period of six (6) seconds.

Cincinnati Bell Serving Area				
Term	1-2499 MOU	2500-4999 MOU	5000+ MOU	
12 Month	\$0.085	\$0.079	\$0.075	
24 Month	\$0.085	\$0.079	\$0.075	
36 Month	\$0.085	\$0.079	\$0.075	

-Calling Cards and International calls are at the regular tariffed rates.

-This Service is billed in six (6) seconds increments, with an initial minimum billing period of six (6) seconds.

Other Charges: (Ameritech and Cincinnati Bell Serving Area)

Nonrecurring Installation Charger First Line	\$50.00
Each Additional Line	\$25.00
Early Termination Penalty	\$100.00/Line

(N)

Issued: January 28, 2000

Effective: March 1, 2000

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No.

Issued by: Amy Hartzler Director, Regulatory and Government Affairs 161 Inverness Drive West Englewood, Colorado 80112 (N)

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)

10.19 Optional Calling Plans to ICG LOCAL service

10.19.1 <u>TOLL CALLING PLANS</u>

A)General

This plan enables Toll Free Service for intraLATA and interLATA and Interstate (outbound and inbound) calling and is available where facilities and billing capabilities permit. All rate plans listed contain Interstate minutes only. No Intrastate minutes are included and all Intrastate minutes will be billed at the rate listed below:

<u>Rate Plans</u>	<u>Monthly Charge</u>	<u>Rate Per Minute</u>
Basic Plan –	\$0.00	\$0.049
Rate Plan 1000 –	\$45.00	\$0.049
Rate Plan 5000	\$175.00	\$0.049
Rate Plan 15,000 –	\$500.00	\$0.049

B) Inbound calls are billed for at a rate of \$0.0147 for the initial 18 seconds and \$0.0049 for each additional 6 seconds.

C) Outbound calls are billed in 6-second increments at a rate of \$.0049 for each 6 seconds.

D)Prices and Charges - Eligible Toll Free calls will be rated, using the prices above, all day, seven days a week.

(N)

Issued: March 9, 2004 Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: Scott Beer Director, Industry &Corporate Affairs 161 Inverness Drive West Englewood, Colorado 80112

(N)

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)

10.19 Optional Calling Plans to ICG LOCAL service (cont'd) 10.19.1 TOLL CALLING PLANS (cont'd)

B. Add-on Credit Card calling Plan 1

1. General

Customers of ICG Business Telecommunications Services that subscribe to this plan must have ICG LOCAL service. Customers must enroll in this plan by completing and returning an enrollment form provided by ICG, by calling an 800 number designated by ICG for this plan, or by enrolling during a marketing contact with ICG.

This plan is an add-on to ICG's local calling and adds, intraLATA and interLATA and Interstate calling and is available where facilities and billing capabilities permit.

2. Eligible Calls — Eligible calls under this plan are: ICG Customer Dialed/Automated card calls and calls placed via (specific numbers so designated by ICG) billed to the customer's ICG calling card associated with their ICG Local, intraLATA and interLATA main billed account

<u>Class</u>	Price Per Minute	Service Charge
Eligible Calling		
Card Calls	\$0.35	\$0.00*

*An additional \$0.30 service charge per call will apply to all calls made from a payphone.

3. Prices and Charges - Eligible Dial Station calls will be rated, using the prices above, all day, seven days a week.

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)

10.19 Optional Calling Plans to ICG LOCAL service (cont'd)

10.19.1 TOLL CALLING PLANS (cont'd)

C. Add on Credit Card calling Plan 2

1. General

Customers of ICG Business Telecommunications Services that subscribe to this plan must have ICG LOCAL service. Customers must enroll in this plan by completing and returning an enrollment form provided by ICG, by calling an 800 number designated by ICG for this plan, or by enrolling during a marketing contact with ICG.

This plan is an add-on to ICG's local calling and adds, intraLATA and interLATA and Interstate calling and is available where facilities and billing capabilities permit.

2. Eligible Calls — Eligible calls under this plan are: ICG Customer Dialed/Automated card calls and calls placed via (specific numbers so designated by ICG) billed to the customer's ICG calling card associated with their ICG Local, intraLATA and interLATA main billed account.

Per call price -		
<u>Class</u>	Price Per Minute	Service Charge
Eligible Calling		
Card Calls	\$0.25	\$0.95*

*An additional \$0.30 service charge per call will apply to all calls made from a payphone

3. Prices and Charges - Eligible Dial Station calls will be rated, using the prices above, all day, seven days a week.

Effective: April 8, 2004

(N)

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)

10.19 Optional Calling Plans to ICG LOCAL service (cont'd)

10.19.1 TOLL CALLING PLANS (cont'd)

D. Add on Toll Free Service Calling Plans

1. General –

Dial Around Service - Dial around rates apply to those instances where a Customer has not selected ICG as their long distance provider, but utilizes the ICG Long Distance Network by dialing ICG's CIC during to the outbound dialing sequence.

Class	Price Per Minute	Service Charge
Eligible Calling Dial Around Calls	\$0.25	\$0.00

Issued: March 9, 2004 Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: Scott Beer Director, Industry &Corporate Affairs 161 Inverness Drive West Englewood, Colorado 80112

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)

10.19 Optional Calling Plans to ICG LOCAL service (cont'd)

10.19.1 TOLL CALLING PLANS (cont'd)

E. Add on Toll Free Service Calling Plans

This plan enables Toll Free Service for intraLATA and interLATA and Interstate (outbound and inbound) calling and is available where facilities and billing capabilities permit. All rate plans listed contain Interstate minutes only. No Intrastate minutes are included and all Intrastate minutes will be billed at the rate listed below:

Rate Plans	Monthly Charge	Rate Per Minute \$0.040	
Basic Dedicated Plan –	\$0.00		
Rate Plan 100,000 –	\$3,000.00	\$0.040	
Rate Plan 500,000 –	\$14,000.00	\$0.040	
Rate Plan 1,000,000	\$25,000.00	\$0.040	

Inbound calls are billed for at a rate of \$0.0120 for the initial 18 seconds and \$0.004 for each additional 6 seconds.

Outbound calls are billed in 6-second increments at a rate of \$.004 for each 6 seconds.

Prices and Charges - Eligible Toll Free calls will be rated, using the prices above, all day, seven days a week.

(Ń)

(N)

10.0 <u>CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)</u> 10.19 Optional Calling Plans to ICG LOCAL service (cont'd)

10.19.1 TOLL CALLING PLANS (cont'd)

F) Other charges

Other charges – In addition to usage charges provision of outbound numbers, directory listings and customized project accounting code charges are listed below.

Feature	Monthly Charge	Non Recurring Charge
Toll Free Number (1-999)	\$2.50	N/A
Toll Free Number (1,000+)	\$1.25	N/A
Directory Listing	\$15.00	\$15.00
Project Accounting Codes -		
Unverified/unforced	\$15.00	\$10.00

Issued: March 9, 2004 Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: Scott Beer Director, Industry &Corporate Affairs 161 Inverness Drive West Englewood, Colorado 80112

10.0 CURRENT RATES AND CHARGES NETWORK SERVICES - RETAIL (PRICE LIST)

10.19 Optional Calling Plans to ICG LOCAL service (cont'd)

10.19.1 <u>TOLL CALLING PLANS</u> (cont'd)

F) Miscellaneous Charges

The Subscriber Line Charge (SLC) is an allowed fee to recover some of the costs associated with providing telephone service. It is not a tax or a fee charged by the government. The Subscriber Line Charges are listed below.

Type of service	Number of lines	Monthly SLC Rate per line
Business lines		
	1-2 Lines	\$6.00
······································	3-12 Lines	\$5.12
	13-23 Lines	\$5.12
·····	24+ Lines	\$1.25
Digital Trunks		
<u> </u>	1-96 Lines	\$3.07
	97-480 Lines	\$3.07
	481+ Lines	ICB

(I) (I)

(I) (I)

Type of service	Number of PRI	Monthly SLC Rate per PRI
PRI		
	1-27 PRI	\$36.00
	28+ PRI	ICB

(D) (N)

(N)

<u>11.0</u> PROMOTIONAL OFFERINGS

11.1 Long Distance Promotion

From September 1, 2000 until October 15, 2000, customers who subscribe to ICG long distance service will be given a \$50.00 credit towards their long distance usage for the first three months of ICG service. This credit will be given at the account level. Customers must subscribe to a minimum one year term in order to receive the credits. Existing ICG customers who do not currently subscribe to ICG long distance service are also eligible for this promotion. Early termination penalties apply as outlined in this tariff.

11.2 Grand Opening

From July 9, through July 31, 1997, for retail business Customers that switch their Local Exchange Telecommunications Service to the Company from the incumbent provider, the Company will waive Non-Recurring Charges, as well as Service order and Change Charges, specified in Sections 6.3 through 6.16 of this PUCO Tariff No. 2. To be eligible for this promotion, the Customer must agree in writing to remain a Customer of the Company's Business Local Exchange Telecommunications Service (which include; Standard Business Line, Single and Multiple Analog PBX Trunk, Direct Inward Dial (DID), Digital Voice Grade DS-1 Trunk Service, Integrated Services Digital Network (ISDN) Primary Rate Interface (PRI), Local Calling Service and Message Telecommunications Service for a minimum term of one year. The Company retains the right to limit the number of DID numbers ordered by a Customer as specified in Section 6.5 of this PUCO Tariff No. 2 to conserve this scarce resource pending the availability of permanent number portability. If the Customer terminates any service on which such non-recurring charges have been waived for any reason prior to the expiration of such one year period, the Company shall charge the Customer in full for each of the non-recurring charges that has previously been waived.

Issued: September 1, 2000

Effective: September 1, 2000

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF

11.3 "Satisfaction Guaranteed" Promotion

From September 1, 1999 and continuing for 180 days thereafter, the company will offer a "Satisfaction Guaranteed" promotion. If the Customer is not fully satisfied with the Company's service within the first ninety days after the Company activates service to the Customer, and if the Customer wishes to return to the Customer's previous local exchange service provider, the Company will pay non-recurring charges the Customer may incur for restoration of the Customer's previous local exchange (dial tone) service, excluding any extra upgrades and features, from the Customer's previous local exchange provider, and excluding any upgrades and additional features provided by the Company to the Customer. The Company's obligation under this Promotion to reimburse the Customer for non-recurring charges shall not exceed \$500.00 per Customer under any circumstances, and further provided that the Customer may not avail of the "Satisfaction Guaranteed" Promotion more than once during the period of this Promotion. Additionally if the Customer's telephone numbers were changed upon the Company's activation of service, the Company is not obligated to restore local exchange service under the previous telephone numbers. This "Satisfaction Guaranteed" Promotion is available to all Customers whose Service from the Company was activated within 90 days prior to the effective date of this tariff filing and to all Customers whose Service from the Company is activated after the effective date of this tariff filing.

11.4 Digital Access Services Promotion

From December 20, 2001 to February 15, 2002, new and existing high volume customers who order "Plan A" Digital Access Services with a 12 month term agreement will receive the following pricing:

Month 1-6 \$275.00 per month, per PRI Months 7-12 \$475.00 per month, per PRI

New customers must order at least 200 PRI in order to receive this promotion. Existing customers must have at least 200 PRI currently in service with ICG and may order new additional circuits under this promotion. This promotion does not apply to any existing PRI services with ICG; existing agreements cannot be renewed in order to receive this promotion. There will be no installation charges applied for circuits installed under this promotion. All other features and Plans of Digital Access Service can be ordered under the regular tariffed rates. Early termination penalties will apply for disconnections prior to the end of the one-year term agreement.

(N)

(N)

Issued: December 20, 2001

Effective: December 20, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Issued by: Amy Hartzler Director, Government & External Affairs 161 Inverness Drive West Englewood, Colorado 80112

11.5 Digital Access Service Three for Three Promotion

Beginning April 10, 2001 through July 1, 2001, for new Digital Access Service (DAS) customers who sign (T) a minimum 36 month service agreement for Plan A or Plan B DAS, a \$1000.00 credit per DAS circuit will be applied to the 12th, 24th and 36th month invoice, for a total credit of \$3,000.00. In order to qualify for the promotion, customers must be located in an on-net building and service must be installed within 60 days of signing the service agreement. Existing customer who add new DAS service and sign a new 36 month contract will qualify for the promotion for the new DAS service only. There will be no credits applied for existing DAS service. Customers who terminate prior to the end of the term agreement, will not receive any further credits and early termination penalties per the terms of this tariff may also apply. Digital Access Three for Three Promotion does not apply for Cincinnati Bell Serving areas.

Issued: April 17, 2001

Effective: April 17, 2001

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF

11.6 **Installation Credit Promotion**

Beginning June 22, 2000 through October 1, 2000, for new customers of the eligible services listed herein, (T) that switch their Local Exchange Telecommunications Service to the Company from an ILEC, and for existing Customers that add one or more of the following eligible services to their existing service, the Company will offer Non-recurring Installation credits pursuant to this Installation Credit Promotion and the particular tariffed services ordered by the customer. This promotion applies to the following eligible services: Standard Single/Multiple Business Line, ICG Standard Business Line Plus, Single/Multiple Analog PBX Trunk, Digital Voice Grade DS-1/Digital Trunks, and Digital Access Service (DAS). Other than the Installation Credit described in this section, all other recurring and non-recurring charges listed in this tariff as applicable to the eligible services continue to apply.

In order to qualify for this Installation Credit promotion, the customer must maintain a minimum of 10 lines or trunks with the Company for the duration of 90 days from date of contract signature. If the Customer does not maintain at least 10 active lines or trunks in service for at least 90 days after the contract signature date, the Company will bill to the Customer and the Customer shall be responsible for payment of the full amount of the credit that was previously extended to the Customer by the Company under this "Installation Credit". New Service Contracts must be signed and dated by both ICG Telecom Group and the Customer by October 1, 2000 and all services must be installed by December 1, 2000 in order to be eligible for this (T) promotion.

This is to attest that ICG will offer this promotion to eligible customers on a non-discriminatory basis.

Issued: June 22, 2000

Effective: June 22, 2000

(T)

Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

11.7 Digital Access Services Promotion Plan B

Beginning April 5, 2002 through December 31, 2002 new "Plan B" Digital Access Service/PRI (Two way or Outbound Only) customers that purchase a full T-1 (24 channels) with a term agreement of 12, 24, 36, 48 or 60 months and current customers that sign a new term agreement of 12, 24, 36, 48 or 60 months will receive the following promotional pricing:

Ameritech Serving Area			Cincinnati Bell Serving Area		
Term	Digital Digital Dig		Excluding Digital Access Loop	Including Digital Access Loop	
12 Month	\$345.00	\$545.00	\$425.00	\$575.00	
24 Month	\$325.00	\$530.00	\$400.00	\$550.00	
36 Month	\$300.00	\$515.00	\$375.00	\$525.00	
48 Month	\$260.00	\$500.00	\$350.00	\$500.00	
60 Month	\$235.00	\$480.00	\$330.00	\$480.00	

This promotion is available exclusively to customers signing a new contract or existing customers renewing service beginning the effective date of this tariff change. Early termination liability charges per the terms of this tariff may also apply for any terminations prior to the end of the term agreement. This promotion does not apply to resale orders and orders involving the use of hybrid circuits that require the transport of traffic at distances greater than 8 miles.

Issued: April 5, 2002

Effective: April 5, 2002

11.7 Digital Access Service - Plan A Promotion

Beginning May 9, 2002 through December 31, 2002, new "Plan A" (inbound only) Digital Access Service/PRI customers that purchase a full T-1 (24 channels) with a term agreement of 12, 24 or 36 months, will receive the following promotional pricing.

Cincinnati Bell Serving Area PRI Plan A - Port Pricing					
		1-5 PRI	6-28 PRI	29-56 PRI	57+ PRI
Term	NRC/PRI	MRC	MRC	MRC	MRC
12 Months	\$300	\$399	\$391	\$383	\$374
24 Months	\$200	\$369	\$361	\$353	\$344
36 Months	\$0	\$339	\$331	\$323	\$314

Ameritech Serving Area PRI Plan A - Port Pricing					
		1-5 PRI	6-28 PRI	29-56 PRI	57+ PRI
Term	NRC/PRI	MRC	MRC	MRC	MRC
12 Months	\$300	\$414	\$403	\$392	\$381
24 Months	\$200	\$369	\$359	\$348	\$337
36 Months	\$0	\$349	\$340	\$329	\$318

Cincinnati Bell & Ameritech Serving Areas PRI Loop Pricing				
On-Net or Hybrid Access Loop	DS1 MRC	DS3 MRC	DS1 NRC	DS3 NRC
< 8 Miles	\$150	\$1,500	\$250	\$500
> 8 Miles	ICB	ICB	ICB	ICB
Cross-Connect	\$99	\$490	\$250	\$500
DS3 Mux	Not Applicable	\$500	Not Applicable	\$0

This promotion is available exclusively to, 1) new customers signing a new contract, 2) existing customers renewing all their ICG Plan A PRI service and, 3) existing customers that only want to sign a new contract term for new PRI service, beginning the effective date of this tariff change. This Plan A PRI Port pricing is the same for all configurations, whether Option 1, 2 or 3. Early termination liability charges per the terms of this tariff apply for any terminations prior to the end of the contract. This promotion does not apply to resale orders.ICG Telecom Group, Inc.

Issued: May 9, 2002

Effective: May 9, 2002

11.8 PRI Promotion

New customers ordering ICG's Two-way PRI and subscribing to Interstate Long Distance between the effective date and June 30, 2004 will receive the following promotional offers based on their contract term:

TERM	MRC Discount	NRC
For a new 1 Year contract	1 month free service	\$500.00 per Two way PRI circuit
For a new 2 Year contract	2 months free service	\$250.00 per Two way PRI circuit
For a new 3 Year contract	3 months free service	\$ 0.00 per Two way PRI circuit

Issued: April 21, 2004

Effective: April 21, 2004

12.0 CURRENT RATES AND CHARGES CARRIER ACCESS SERVICE (PRICE LIST)

12.1 <u>Rates and Charges, for Customer's Using Company's Local Exchange Telecommunications</u> Service

The following rates and charges are applicable to the situation where the Company is providing Local Exchange Telecommunications Service to its Customer and the Company's switch is providing dial tone to the Customer.

12.1.1 Within Ameritech's Service Territory

12.1.1.1	Entrance Facility		
	Per DS1	\$151.29 per month	
	Per DS3	\$2,000.00 per month	
	Multiplexing (DS3/DS1 multiplexing)	\$280.50 per month	
12.1.1.2	Local Transport		
	DS1 DTT (per termination)	\$82.00 per month	
	DS1 DTT (per mile)	\$23.38 per mile per month	
	DS3 DTT (per termination)	\$399.50 per month	
	DS3 DTT (per mile)	\$124.00 per mile	
	Per MOU per termination	\$0.00020376	(R)

Issued: November 29, 2002

Effective: December 29, 2002

10.0

.....

12.0	CURREN	<u>FRATES AND CHARGES C</u>	CARRIER ACCESS SERVICE (PRICE LIST) (co	ont'd)
12.1	<u>Rates and</u> <u>Service</u> (co	-	ng Company's Local Exchange Telecommunica	<u>tions</u>
	12.1.1.3	Access Tandem		
		Per MOU	\$0.000520	(R)
	12.1.1.4	End Office		
		Per MOU	\$0.01131408	(D)
	12.1.1.5	800 Database Query with	out Pots Translation	(R)
		Per Query	\$0.002314	
12.2	Nonrecurr	ing Charges		
	12.2.1	Installation Charges		
			1	1

CURRENT AND AND CHARGES CARDIDE A CO

Terminating Connection, Per DS1, Per Serving Wirwo	Way Originating and Terminating or One-Way
Center	Originating Connection, Per DS1, per Serving
	Wire Center or End Office
\$500.00	\$1,400.00

The Nonrecurring Charges for installation of DS-3 service will be established on an Individual Case Basis ("ICB").

The charge for expediting a service installation interval will be established on an Individual Case Basis ("ICB").

Issued: December 2, 2002

Effective: January 1, 2003

<u>14.0</u> **TOLL CALLING PLANS**

14.1 Add on Direct Dialed Long Distance Calling Plans

14.1.1 General

A) This plan enables intraLATA and interLATA and Interstate (outbound and inbound) calling and is available where facilities and billing capabilities permit. All rate plans listed contain Interstate minutes only. No Intrastate minutes are included and all Intrastate minutes will be billed within the rate listed below:

[Monthly	Per Minute	Initial 18	6 Second
	Charge	Rate	Second	Incremental
			Inbound	Rate
			Rate	
Basic Plan –				
Minimum	\$0.00	\$0.007	\$0.0021	\$0.0007
Maximum	\$4.00	\$0.250	\$0.0750	\$0.0250
Rate Plan 1,000 –				
Minimum	\$15.00	\$0.007	\$0.0021	\$0.0007
Maximum	\$60.00	\$0.250	\$0.0750	\$0.0250
Rate Plan 5,000 -				
Minimum	\$100.00	\$0.007	\$0.0021	\$0.0007
Maximum	\$240.00	\$0.250	\$0.0750	\$0.0250
Rate Plan 15,000				
Minimum	\$300.00	\$0.007	\$0.0021	\$0.0007
Maximum	\$700.00	\$0.250	\$0.0750	\$0.0250

- B) Inbound calls are billed for at the initial 18 second inbound rates and at the 6 second incremental rate for each additional 6 seconds.
- Outbound calls are billed in 6-second increments at the 6-second increments al rate for C) each 6 seconds period.
- D) Prices and Charges - Eligible Toll Free calls will be rated, using the prices above, all day, seven days a week.

Effective: November 10, 2003

Issued by: Scott Beer Director, Industry & Corporate Affairs 161 Inverness Drive West Englewood, Colorado 80112

(N)

(N)

(N)

(N)

<u>14.0</u> <u>TOLL CALLING PLANS</u> (cont'd)

14.1 Add on Direct Dialed Long Distance Calling Plans (cont'd)

14.1.2 Add on Credit Card calling Plan 1

A) General

Customers of ICG Business Telecommunications Services that subscribe to this plan must have ICG LOCAL service. Customers must enroll in this plan by completing and returning an enrollment form provided by ICG, by calling an 800 number designated by ICG for this plan, or by enrolling during a marketing contact with ICG.

This plan is an add-on to ICG's local calling and adds, intraLATA and interLATA and Interstate calling and is available where facilities and billing capabilities permit.

B) Eligible Calls

Eligible calls under this plan are: ICG Customer Dialed/Automated card calls and calls placed via (specific numbers so designated by ICG) billed to the customer's ICG calling card associated with their ICG Local, intraLATA and interLATA main billed account.

<u>Class</u>	Price Per Minute	Service Charge
Eligible Calling		
Card Calls		
Minimum	\$0.03	\$0.00*
Maximum	\$0.50	\$1.50*

*Payphone service charge \$0.30 per call

C) Prices and Charges

Eligible Dial Station calls will be rated, using the prices above, all day, seven days a week.

Effective: November 10, 2003

14.0 TOLL CALLING PLANS (cont'd)

<u>14.1</u> Add on Direct Dialed Long Distance Calling Plans (cont'd)

- 14.1.3 Add on Credit Card calling Plan 2
 - A) General

Customers of ICG Business Telecommunications Services that subscribe to this plan must have ICG LOCAL service. Customers must enroll in this plan by completing and returning an enrollment form provided by ICG, by calling an 800 number designated by ICG for this plan, or by enrolling during a marketing contact with ICG.

This plan is an add-on to ICG's local calling and adds, intraLATA and interLATA and Interstate calling and is available where facilities and billing capabilities permit.

B) Eligible Calls -- Eligible calls under this plan are: ICG Customer Dialed/Automated card calls and calls placed via (specific numbers so designated by ICG) billed to the customer's ICG calling card associated with their ICG Local, intraLATA and interLATA main billed account.

Per call price <u>Class</u> Price Per Minute Service Charge
Eligible Calling
Card Calls
Minimum \$0.00 \$0.00*
Maximum \$2.50 \$2.50*

*An additional \$0.30 service charge per call will apply to all calls made from a payphone.

C) Prices and Charges - Eligible Dial Station calls will be rated, using the prices above, all day, seven days a week.

(N)

Issued: October 9, 2003 Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF. Effective: November 10, 2003

14.0 TOLL CALLING PLANS (cont'd)

14.1 Add on Direct Dialed Long Distance Calling Plans (cont'd)

14.1.4 Add on Toll Free Service Calling Plans

A) General -

Dial Around Service - Dial around rates apply to those instances where a Customer has not selected ICG as their long distance provider, but utilizes the ICG Long Distance Network by dialing ICG's CIC during to the outbound dialing sequence.

<u>Class</u> Eligible Calling	Price Per Minute	Service Charge
Dial Around Calls		
Minimum	\$0.03	\$0.00
Maximum	\$0.95	\$2.00

(N)

Issued: October 9, 2003 Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF. Effective: November 10, 2003

(N)

14.0 TOLL CALLING PLANS (cont'd)

14.1 Add on Direct Dialed Long Distance Calling Plans (cont'd)

14.1.5 Add on Toll Free Service Calling Plans

A) General

This plan enables intraLATA and interLATA and Interstate (outbound and inbound) calling and is available where facilities and billing capabilities permit. Plans do not include Intrastate minutes. All Intrastate minutes will be billed within the rates listed below:

	Monthly	Per Minute	Initial 18	6 Second
	Charge	Rate	Second	Incremental
			Inbound	Rate
			Rate	
Basic Dedicated Plan –				
Minimum	\$0.00	\$0.007	\$0.0021	\$0.0007
Maximum	\$25.00	\$0.250	\$0.0750	\$0.0250
Rate Plan 100,000 –	<u> </u>			
Minimum	\$700.00	\$0.007	\$0.0021	\$0.0007
Maximum	\$6,000.00	\$0.250	\$0.0750	\$0.0250
Rate Plan 500,000 -				
Minimum	\$6,000.00	\$0.007	\$0.0021	\$0.0007
Maximum	\$25,000.00	\$0.250	\$0.0750	\$0.0250
Rate Plan 1,000,000 –				
Minimum	\$12,000.00	\$0.007	\$0.0021	\$0.0007
Maximum	\$50,000.00	\$0.250	\$0.0750	\$0.0250

- B) Inbound calls are billed for at the initial 18 second inbound rates and at the 6 second incremental rate for each additional 6 seconds.
- C) Outbound calls are billed in 6-second increments at the 6-second increments al rate for each 6 seconds period.
- D) Prices and Charges Eligible Toll Free calls will be rated, using the prices above, all day, seven days a week.

Effective: November 10, 2003

14.0 TOLL CALLING PLANS (cont'd)

14.1 Add on Direct Dialed Long Distance Calling Plans (cont'd)

14.1.6 Other charges

A) General

Other charges – In addition to usage charges provision of outbound numbers, directory listings and customized project accounting code charges are listed below.

Feature	Monthly Charge	Non Recurring Charge
Toll Free Number (1-999)		Churge
Minimum	\$1.00	N/A
Maximum	\$5.00	N/A
Toll Free Number (1,000+)		
Minimum	\$0.50	N/A
Maximum	\$4.00	N/A
Directory Listing		
Minimum	\$2.50	\$2.50
Maximum	\$25.00	\$30.00
Project Accounting Codes –	···· ···	
Unverified/unforced		
Minimum	\$2.50	\$2.50
Maximum	\$25.00	\$25.00

Issued: October 9, 2003 Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF.

Effective: November 10, 2003

Issued by: Scott Beer Director, Industry &Corporate Affairs 161 Inverness Drive West Englewood, Colorado 80112 (N)

(N)

(N)

(N)

14.0 TOLL CALLING PLANS (cont'd)

14.1 Add on Direct Dialed Long Distance Calling Plans (cont'd)

14.1.7 Miscellaneous Charges

A) General

The Subscriber Line Charge (SLC) is an allowed fee to recover some of the costs associated with providing telephone service. It is not a tax or a fee charged by the government. The Subscriber Line Charges are listed below.

Type of Service	Number of lines	Monthly SLC	Monthly SLC
		Rate per line	Rate per line
Business lines		Minimum	Maximum
	1-2 Lines	\$3.00	\$12.00
	3-12 Lines	\$2.00	\$10.00
	13-23 Lines	\$1.00	\$8.00
	24 + Lines	\$.75	\$7.00
Digital Trunks			
	1-96 Lines	\$1.00	\$5.50
	97-480 Lines	\$0.50	\$4.00
	481+ Lines	ICB	ICB

Type of service	Number of PRI	Monthly SLC	Monthly SLC
		Rate per PRI	Rate per PRI
PRI		Minimum	Maximum
	1-27 PRI	\$25.00	\$50.00
	28+ PRI	ICB	ICB

Issued: October 9, 2003 Filed under authority of Order of the Public Utilities Commission of Ohio, in Case No. 90-9004-TP-TRF. Effective: November 10, 2003

This foregoing document was electronically filed with the Public Utilities

Commission of Ohio Docketing Information System on

7/1/2008 5:53:51 PM

in

Case No(s). 08-0840-TP-ATA

Summary: Tariff Exhibits A & B for the Application of ICG Telecom Group, Inc. to Detariff Certain Tier 2 Services in its PUCO Tariff No. 2 electronically filed by Karen M Hyde on behalf of ICG Telecom Group, Inc.